



A Comprehensive Review of Sizing and Energy Management Strategies for Optimal Planning of Microgrids with PV and Other Renewable Integration

Fadi Agha Kassab 🗅, Rusber Rodriguez 🗅, Berk Celik 🗅, Fabrice Locment 🗅 and Manuela Sechilariu *🗅

Avenues, Centre Pierre Guillaumat, Université de Technologie de Compiègne, CS 60 319, 60203 Compiègne, France; fadi.agha-kassab@utc.fr (F.A.K.); rusber-octavio.rodriguez-velasquez@utc.fr (R.R.); berk.celik@utc.fr (B.C.); fabrice.locment@utc.fr (F.L.)

* Correspondence: manuela.sechilariu@utc.fr; Tel.: +33-03-4423-7317

Abstract: This article comprehensively reviews strategies for optimal microgrid planning, focusing on integrating renewable energy sources. The study explores heuristic, mathematical, and hybrid methods for microgrid sizing and optimization-based energy management approaches, addressing the need for detailed energy planning and seamless integration between these stages. Key findings emphasize the importance of optimal sizing to minimize costs and reduce carbon dioxide (CO₂) emissions while ensuring system reliability. In a pedagogical manner, this review highlights the integrated methodologies that simultaneously address sizing and energy management and the potential of emerging technologies, such as smart grids and electric vehicles, to enhance energy efficiency and sustainability. This study outlines the importance of accurate load modeling and carefully selecting models for renewable energy sources and energy storage systems, including degradation models, to achieve long-term operational efficiency and sustainability in microgrid design and operation. Future research should focus on developing multi-objective optimization techniques and incorporating cutting-edge technologies for improved microgrid planning and operation.



1. Introduction

Microgrids (MGs) are distributed energy systems that can operate autonomously or be interconnected to the primary power grid, efficiently managing energy generation, storage, and consumption within a defined electrical community [1,2]. These local grids could integrate diverse distributed energy resources (DER), including photovoltaic (PV) systems, wind turbines (WTs), and small hydroelectric power plants. Integrating renewable energy sources is a crucial feature of microgrids, introducing specific challenges due to the intermittency of these sources [3,4]. A microgrid design typically considers minimizing the loss of power supply probability (LPSP) [5]. Furthermore, microgrids could integrate load-shedding schemes, where non-critical loads can be selectively disconnected when the demand exceeds the supply capacity [6,7]. Energy management is crucial in microgrid operation to meet energy demands appropriately. It refers to controlling and optimizing energy generation, storage, and consumption to meet the community's needs. Therefore, detailed and focused energy management, coupled with an adequate energy storage system (ESS), is critical to the successful operation of microgrids, especially in non-interconnected regions where reliability and autonomy are critical. Given the complexity and importance of these systems, it is essential to pay close attention to the design and operation of a microgrid. One of the primary stages in this process is energy planning, which includes selecting energy sources and sizing the sources chosen as a core step [8]. It also involves strategically



Citation: Agha Kassab, F.; Rodriguez, R.; Celik, B.; Locment, F.; Sechilariu, M. A Comprehensive Review of Sizing and Energy Management Strategies for Optimal Planning of Microgrids with PV and Other Renewable Integration. *Appl. Sci.* 2024, *14*, 10479. https://doi.org/ 10.3390/app142210479

Academic Editor: Alessandro Lo Schiavo

Received: 3 October 2024 Revised: 8 November 2024 Accepted: 11 November 2024 Published: 14 November 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). configuring the microgrid components and balancing energy production and consumption to minimize initial investments, operating costs, maintenance, and replacement costs [9].

Microgrid energy planning has gained considerable attention in the scientific community and among power system stakeholders due to its potential beneficial impacts on efficiency, emissions, and carbon footprint reduction. It is a crucial element in the design of sustainable energy infrastructures [10]. Numerous strategies have been developed for microgrid energy planning, which is helpful but also needs to be clarified, as the variety of methods makes it difficult to determine a suitable method for each situation. The choice of energy planning methodology depends mainly on the design objectives and the specific needs being addressed. Complex methodologies could consider much information to provide exact results for particular cases. For example, Figure 1 presents a breakdown of the data used in [11] for the energy planning of a microgrid, where a detailed scan of a remote rural region was performed. The region was divided into small subareas; each matrix position represented the subarea's latitude and longitude location. The hypermatrix was composed of layers recording the characteristics of each point, such as the location of houses, terrain altitude, solar potential, wind potential, hydraulic potential, and ecologically protected zones. This information matrix was used to develop the energy planning of an isolated microgrid to power the houses enclosed in the study region. This type of study typically generates high-accuracy results. However, they entail a high degree complexity and require a lot of resources to survey the data and reach an optimal solution. The energy planning of a microgrid generally involves these steps: (i) the selection of energy sources, (ii) the sizing of these sources, and (iii) the definition of the energy management strategy. The level of detail in each phase might vary depending on the design objective [12]. Researchers have developed methodologies that combine these steps in different ways, either sequentially, simultaneously, or in a hybrid approach. Figure 2 shows the basic steps of microgrid energy planning and the order in which they could be addressed sequentially or simultaneously.



Figure 1. Overview of information processing in [11] for accurate energy planning of an isolated rural microgrid. (**a**) Division of the study region into subareas; (**b**) Layers recording the characteristics of the subareas.

Given the importance of guiding energy planning methodologies, several researchers have conducted reviews in this area. One aspect that has attracted the most attention is the definition of an energy management strategy (EMS), whether at the design or microgrid operation stage. For example, Liu et al. [13] developed a comprehensive review on energy management in microgrids with ESS. This review discusses recent advances in microgrid architecture, energy management models, solution methodologies, and future research directions. The microgrid architecture is classified into alternating current (AC), direct current (DC), hybrid AC/DC, and multi-energy microgrids. Each type is analyzed in terms of efficiency, flexibility, and resilience and their ability to integrate ESS and renewable sources. This research also evaluates different approaches to energy management in microgrids, including mathematical programming, adaptive dynamic programming, and deep reinforcement learning. It covers a wide range of ESS technologies, highlighting their crucial role in the reliable and stable operation of microgrids in grid-connected and

islanded modes. It identifies gaps in problem formulation, especially in low carbon and resilience management applications, and suggests the need for more efficient computational solutions considering information security. It also highlights the need for further resilience management and information security research. In the same regard, Abdi [14] conducted a review of surveys of microgrid EMS. This review synthesized previous studies, providing an overview of the advances and challenges in microgrid energy management and examining different structures and schemes, including centralized, decentralized, and distributed approaches. It also reviewed optimization techniques used in energy management, such as mathematical programming, heuristic methods, and machine learning models. It also focused on the challenges of integrating renewable energy, cyber security, and microgrid control strategies. This review guides researchers interested in energy management in microgrids, covering aspects of power flow optimization, reliability assessment, and the application of advanced control techniques.



Figure 2. Basic steps in the energy planning of a microgrid.

Meanwhile, Shafiullah et al. [15] focused on analyzing the state-of-the-art of EMS for microgrids. This review covers a comprehensive analysis of EMS, applied optimization methods, and implementation challenges associated with microgrids in grid-connected systems. It discusses management strategies for a microgrid's main components, including charging, generation, and ESS. It reviews optimization approaches, such as classical, metaheuristic, and artificial intelligence-based methods, to improve the operational efficiency of microgrids and reduce costs. This research also addresses the technical, social, economic, and regulatory challenges facing microgrids, highlighting the need for robust regulatory frameworks and policies that facilitate the effective integration of microgrids and deepen the operational efficiency, integration of renewable sources, and resilience of microgrids in future scenarios. Jirdehi et al. [2] conducted a comprehensive review on the management of microgrid components, network structures, operation modes, ESS, load types, problem modeling, resolution procedures, and other relevant aspects from the perspective of microgrid management. They highlighted the importance of optimized energy management and control methods for microgrid operation, addressing energy balance, cost optimization, and energy loss reduction. The review also analyzed the effect on the operation of generators, ESS, and loads, and addressed structural configurations of microgrids, such as AC, DC, and hybrid networks. It explored advanced mathematical modeling techniques

and optimization methods, highlighting their application in managing and controlling microgrids under uncertainty and varying conditions. A particularly interesting aspect of this review is its focus on the statistical comparison of different methods and approaches used in microgrid management. This comparison provides a clear perspective on current developments and trends in the field, allowing researchers to identify critical areas for future research.

Similarly, Ahmad et al. [16] presented a comprehensive review of microgrids' energy management and control strategies. This review analyzed the methodologies and techniques employed for microgrid energy management and control optimization, focusing on recent advances and future challenges. It examined various optimization techniques, from mathematical methods to metaheuristic algorithms, used to maximize operational efficiency and minimize costs in microgrid energy management. It also reviewed real-time control strategies for maintaining system stability, including voltage, frequency, and power control in different microgrid operating scenarios. A distinctive aspect of this review is its focus on the connection between theory and practice. It discusses the influence of IEEE standards 1547 [17], 1547.3 [18], 1547.6 [19], and 2030 [20] on implementing management and control strategies in microgrids. Battula et al. [21] also developed an outstanding review on EMS approaches in microgrids, highlighting classical, heuristic, and intelligent methods to optimize operation and efficiency. They examined energy management methodologies, including classical mathematical programming, metaheuristic algorithms, and artificial intelligence-based techniques. Each approach was discussed in terms of its applicability and effectiveness in different microgrid scenarios. They delved into methods considering multiple objectives such as cost minimization, reliability improvement, and emission reduction, emphasizing the complexity of managing these factors simultaneously. The review also addressed the ancillary infrastructure needed, such as the Internet of Things and smart meters, to effectively implement management strategies in microgrids.

These outstanding reviews from the scientific community have pointed out the importance of energy management in microgrids. This compilation also reveals the importance of deepening the integration of the sizing and energy management stages and the definition of the objective function for microgrid energy planning. These stages could be integrated differently, giving rise to many variations in microgrid planning methodology. Addressing this concern, this paper develops a detailed review of the most relevant sizing and energy management strategies for microgrid energy planning and how these techniques could be integrated to address specific objectives. It is intended to guide readers on the energy planning strategy most appropriate for their needs and goals. The methodology employed for this review follows a structured approach to ensure comprehensive coverage of the literature. The primary focus was on the energy planning of microgrids integrating renewable energy sources (RES). The steps taken in this literature review were as follows: (i) Literature search. A thorough search of research articles was conducted using search terms such as "microgrid energy planning", "renewable integration in microgrids", "microgrid sizing methodologies", and "energy management in microgrids". (ii) Selection criteria. The articles selected for review were chosen based on their relevance to microgrid energy planning, specifically focusing on integrating RES, sizing methodologies, and energy management strategies. Only peer-reviewed articles, high-quality conference papers, and doctoral theses published in the last decade were considered. (iii) Identification of key methodologies. The main microgrid sizing and energy management methodologies were identified and categorized. This involved analyzing the approaches used in various studies and their effectiveness in addressing the challenges associated with microgrid planning. (iv) Comparison of sequential and simultaneous optimization. The review also contrasted methodologies that employ sequential optimization of sizing and energy management with those that utilize simultaneous optimization. This comparison aimed to highlight the advantages and disadvantages of each approach.

This review comprehensively analyzes sizing and energy management strategies tailored to microgrids integrating renewable energy sources. It addresses the gap in the

literature by thoroughly comparing sequential and simultaneous optimization techniques, highlighting their effectiveness in balancing energy production, cost minimization, and system sustainability. Additionally, it discusses integrating emerging technologies, such as electric vehicles and smart grids, which present significant potential for enhancing microgrid efficiency. This study emphasizes methodologies and approaches that support decision-making in energy planning and management under varying conditions, including the intermittency of renewable sources and the complexities of non-interconnected regions. Furthermore, it is intended to serve as a pedagogical resource for researchers and practitioners new to microgrid planning, energy management, and sizing. This paper contributes, in a pedagogical manner, to the existing knowledge of microgrid energy planning; its key contributions are as follows:

- Comprehensive review: The paper thoroughly reviews the current methodologies for microgrid energy planning, highlighting the main strategies for both sizing and energy management.
- Methodological comparison: This paper compares sequential and simultaneous optimization approaches, providing insights into their benefits and challenges.
- Integration strategies: This study discusses various integration strategies for renewable energy sources within microgrids, emphasizing the importance of efficient energy management to address the intermittency of renewables.
- Practical guidance: By synthesizing the findings from numerous studies, this paper
 provides practical advice for researchers and practitioners on selecting the most
 appropriate energy planning strategy based on specific needs and objectives.

The remainder of this review paper is structured in the following sections: Section 2 remarks on the topology of microgrids, including isolated and interconnected systems, and their operational modes in AC and DC configurations. Section 3 reviews different methodologies for microgrid sizing, including heuristic, mathematical, and hybrid approaches. Section 4 explores various energy management approaches within microgrids, highlighting optimization-based methods and commercial software tools. Section 5 discusses optimization approaches in microgrids' sizing and energy management, and analyses their advantages and challenges. Lastly, Section 6 provides concluding remarks and insights on future work relating to this study.

2. Remarks on the Topology of Microgrid

A microgrid is a local energy system integrating distributed generation, energy storage, and controllable loads within a defined electrical network. Microgrids stand out among lowpower generation systems for their ability to operate independently of the primary grid and manage the energy sources that comprise them. Typically, energy management integrates an algorithm to optimize operation [3]. These networks could be classified according to their connection and mode of operation. According to the connection, there are isolated and grid-connected microgrids. Isolated microgrids operate independently of a primary power grid, rendering them suitable for remote areas without access to conventional electrical infrastructure. On the other hand, grid-connected microgrids operate synchronously with a primary power grid, allowing the import and export of energy as needed [12,22]. Microgrids can operate in AC, DC, or a hybrid combination. AC microgrids are the most common due to their compatibility with existing electrical infrastructure and conventional devices. DC microgrids, although less common, are gaining popularity for specific applications. Hybrid configurations combine AC and DC, offering greater flexibility and efficiency by optimizing the operation of different types of loads and generation sources [23]. Microgrids, with their ability to manage generation intermittency, represent a promising solution for integrating RES. Their significant role in improving local energy efficiency offers feasibility for a more sustainable future.

Researchers have suggested categorizations for microgrids given the above scenarios. Figure 3 highlights some commonly used configurations, modes of operation, and applications in this field. The following sections address these classifications. Section 2.1 presents

Classifying Criteria Microgrid type Applications microgrids Isolated *Agriculture Mode of operation • Grid-connected *Commercial • A(✤Industrial • DC Power type Hybrid Institutional MG Military Centralized Decentralized *Municipal *Residencial Single-phase Number of Phases Three-phase Transportation

the alternative microgrid operation modes. Section 2.2 provides comments on integrating RES. Then, Section 2.3 focuses on microgrid applications. Finally, Section 2.4 provides a general framework for the cost of microgrids in planning.

Figure 3. Standard classifications of microgrids in terms of operation.

2.1. Interconnection and Mode of Operation of Microgrids

The topology of a microgrid is essential for determining its efficiency, stability, and ability to integrate with other energy systems. Interconnection and operation mode influence the overall performance of a microgrid. Microgrids can operate in two modes: (i) grid-connected and (ii) isolated. In grid-connected mode, the microgrid is synchronized with the primary electrical grid, allowing bidirectional energy exchange [24,25]. This setup enhances system stability and demand management, as the primary grid can serve as a backup during fluctuations in local generation or demand. However, this mode requires a robust infrastructure and precise coordination to maintain frequency and voltage at the point of common coupling. The microgrid operates independently in isolated mode, crucial in remote areas or during emergencies. This mode presents additional challenges, such as balancing generation and consumption in real time. Integrating ESS, like batteries, is vital to managing the intermittency of renewable sources and ensuring a continuous energy supply [26].

On the other hand, most microgrids operate in AC due to compatibility with traditional electrical systems and the majority of electrical equipment that runs on AC. Once phase and frequency synchronization is maintained, AC microgrids can integrate various generation sources, such as diesel generators, WT, and PV systems [27]. DC microgrids are gaining popularity, especially in applications with critical energy efficiency, such as PV systems and battery storage. DC operation eliminates losses associated with AC-DC conversions, leading to higher energy efficiency. However, integrating with existing grid infrastructure and ensuring system protection pose significant challenges [28]. Hybrid mode combines the advantages of AC and DC systems within a single microgrid. This approach allows greater flexibility in managing diverse generation and storage sources. Hybrid microgrids can leverage the efficiency of DC systems for specific applications, such as electric vehicle charging, while maintaining compatibility with existing AC infrastructure [29]. Table 1 compares microgrids' operation modes and power supply types. It highlights key features, challenges, and typical applications associated with each mode, understanding their advantages and limitations.

Feature	Grid-Connected Mode	Isolated Mode	AC Operation	DC Operation	Hybrid Operation (AC/DC)
Connectivity	Connected to the main grid.	Independent.	Compatible with tradi- tional systems.	Efficient for specific sys- tems.	Integration of AC and DC.
Flexibility	High, but depends on the main grid.	Limited by local ca- pacity.	High for integrating di- verse sources.	Limited by equipment and applications.	High, allows specific use of AC and DC.
Efficiency	Moderate, depends on grid.	Variable, depends on ESS.	Moderate due to con- version losses.	High, eliminates con- version losses.	High, optimizes based on source.
Challenges	Requires synchroniza- tion and stability.	Real-time generation and load manage- ment.	Frequency and phase synchronization.	System protection and conversion.	Integration and man- agement of different standards.
Applications	Urban areas, grid sup- port.	Remote areas, critical sites.	Traditional systems, ex- isting infrastructure.	PV systems, batteries, EVs.	Modern infrastructure, mixed installations.

Table 1. Operating modes and	types of	power sup	pply for	microgrids.
-------------------------------------	----------	-----------	----------	-------------

2.2. Integration of Renewable Energy Sources

Given the increasing focus on RES to achieve sustainability goals, their integration into microgrids should be emphasized. Understanding the challenges and benefits of different types of RES provides the basis for design strategies and optimization methods. This section explores the role of RES in microgrid systems, and the following sections discuss their impact on energy management. RES have been increasingly integrated worldwide in recent years due to their outstanding potential to mitigate climate change, improve energy security, promote economic development, create employment, and offer sustainable access to energy, especially in regions with limited access [30]. Leading RES institutions such as the Renewable Energy Policy Network for the 21st Century (REN21) [30], the International Energy Agency (IEA) [31], and the International Renewable Energy Agency (IRENA) [32] agree with the following definition: renewable energies are those obtained from natural sources that are continuously replenished and could be considered inexhaustible in the human time scale. Their environmental impact is considerably reduced compared to traditional fossil fuels. The following are examples of renewable energy which are detailed in the 2024 REN21 report [30]:

- Solar energy comes from solar radiation that can be converted into electricity or heat. It is one of the most abundant and accessible sources of renewable energy.
- Wind energy generated from the wind moves the blades of wind turbines to produce electricity. It is incredibly efficient in regions with strong and constant winds.
- Hydroelectric power harnesses water movement, generally in rivers or dams, to generate electricity. It is one of the most traditional forms of renewable energy.
- Geothermal energy originates from the natural heat of the Earth's interior. It can be used to generate electricity or for direct heating in geothermal areas.
- Biomass from organic materials such as wood, agricultural residues, and organic waste can generate energy through combustion or conversion into biofuels.

RES have been successfully integrated into microgrids for power generation through various approaches. Their integration is crucial in microgrid planning due to the increasing need for sustainability and carbon emission reduction. They offer significant advantages, such as availability and low environmental impact [33]. However, there are several challenges associated with these sources' intermittent and variable nature. Addressing these challenges requires optimizing the size and combination of these resources, balancing generation and demand, minimizing costs, and maximizing efficiency [34]. It could also involving generating the hybridization of AC and DC operation modes, involving several stages of power conversion, and making an indispensable EMS that guarantees the control of energy sources and the supply to the demand. Figure 4 shows an approach to the connection diagram of a microgrid integrating RES and power conversion stages. This diagram shows that microgrids with RES could also include controllable power sources, such as a diesel generator, to help ensure the reliability of the power system. This figure

also outlines the AC and DC operation of a microgrid. Each side has a central operating bus. The DC bus interconnects sources and loads of a DC nature, even allowing the connection of an electric vehicle (EV) as a flexible load. The AC side interconnects traditional AC loads and is interconnected with the local AC power network. A bi-directional DC/AC converter interconnects the DC and AC sides. The following section discusses general applications of microgrids.



Figure 4. Schematic of a microgrid with renewable energy integration.

2.3. Microgrid Applications

Microgrid applications are diverse, with the most common being rural electrification in remote areas, industrial facilities requiring efficient energy management and resilience, military installations seeking energy independence and security, and urban or commercial environments benefiting from increased energy reliability. They can be implemented in various configurations, isolated from or connected to the main power grid, to address each sector's specific needs and challenges [35]. In rural areas, microgrids provide a viable solution for electrification, reducing dependence on fossil fuels and improving the quality of life. Operating in isolated mode is particularly useful in these areas, where the central grid infrastructure may need to be improved or non-existent [11]. In the industrial sector, microgrids enable more efficient energy management and the integration of DER. By reducing operational costs and enhancing the resilience of the facility's energy system, microgrids contribute significantly to overall efficiency [36]. Microgrids also have applications in the military sector, where energy independence and security are priorities. Operating autonomously and ensuring a continuous energy supply in emergencies is crucial.

The following are outstanding examples of the practical implementation of optimal microgrid planning, which also present challenges inherent to their specific environments. First, the study in [11] describes using an particle swarm optimization (PSO) algorithm for energy planning of an isolated microgrid in Colombia. This system is designed to supply energy to a remote population of seven houses in a region with difficult access and high resource transportation costs. The optimization strategy focused on taking advantage of local natural resources, mainly due to the limited availability of conventional energy sources and logistical difficulties for regular supply. This application illustrates how geographic

and access particularities can condition design decisions in microgrid planning, focusing on sustainability and energy autonomy.

On the other hand, the Gorona del Viento project on the island of El Hierro, Spain, implements a hybrid renewable energy system that combines wind energy with pumped hydroelectric storage. This project has enabled the island to operate nearly self-sufficiently, significantly reducing its dependence on traditional diesel generators. The system design takes advantage of the island's topography, using the La Caldera crater as an upper reservoir for water storage, which facilitates the accumulation of energy during periods of high wind production. However, challenges were faced during implementation, mainly due to the limited storage capacity of the reservoir, which resulted in reduced grid stability during periods of low wind speed [37]. This case highlights the importance of adapting system components to the geological and climatic characteristics of the site and how the combination of energy sources and storage strategies can help mitigate the intermittency of renewable sources.

These examples highlight how the selection of optimization strategies and microgrid configuration can vary according to the local context and available resources, impacting both costs and carbon emission reductions. Table 2 provides an overview of microgrid applications across different sectors.

Table 2. Applications of microgrids in some sectors.

Sector	Applications	Benefits
Rural	Electrification of remote areas.	Reduces dependence on fossil fuels, improves quality of life.
Industrial	Efficient energy management, integration of DER.	Reduces operational costs, increases energy efficiency.
Military	Energy independence, security.	Reliable operation in critical situations, risk reduction.

The topology of microgrids is essential for their efficiency, stability, and sustainability, especially when integrating renewable sources. The choice of operation mode and electrical configuration influences a microgrid's flexibility and technical feasibility and adaptation to various applications, from rural areas to industrial and military environments. The following sections discuss sizing methodologies Moreover, exploring energy management strategies that optimize microgrid operation provides a comprehensive analysis of tools in microgrid energy planning. Optimizing the sizing and operation of microgrids in various applications requires considering the characteristics of each environment and sector, along with the relevant indicators for energy source selection. Further analysis of these factors, including optimal resource use, system resilience, and constraints tailored to different operating conditions, refines energy planning. Grid-interconnected microgrids, for example, improve system resilience and flexibility with star, ring, or mesh configurations, enabling power sharing and improving the reliability of critical loads [38,39].

In remote or resource-constrained environments, measures such as "mass per unit load" (MPUL) provide an indicator of efficiency by assessing system mass relative to power demand, making this measure applicable in regions with high transportation and installation costs [40]. Adequate microgrid protection relies on adaptive strategies, such as directional overcurrent relays and coordinated algorithms, to maintain stability amid fluctuating loads and bidirectional power flows [38]. Advanced battery degradation models further aid in selecting durable energy storage options and optimizing system lifetime under harsh conditions [40]. This planning approach, which integrates adaptive architectures, efficient resource use, and robust protection strategies, improves the performance of microgrids in various application scenarios.

2.4. Cost Framework in Microgrid Energy Planning

Defining the cost function is crucial in microgrid energy planning as it guides the objectives influencing the microgrid's sizing and energy management. Researchers commonly employ indicators such as the levelized cost of energy (LCOE) and the levelized carbon emission (LCE), which provide insight into the cost per unit of energy generated and the

$$C_{annual} = CRF \cdot C_{total} \tag{1}$$

Here, *CRF* is a capital recovery factor, and C_{total} is the total cost over the project lifetime. *CRF* is a financial metric used to determine the annual capital cost from the project lifetime cost [42]. *CRF* and C_{total} are outlined in (2) and (3), respectively.

$$CRF = \frac{d \cdot (1+d)^Q}{(1+d)^Q - 1}$$
(2)

$$C_{total} = C_{inv} + C_{O\&M} + C_{rep} + C_{EG} - C_{sv} + C_{special}$$

$$\tag{3}$$

Here, *d* is the annual rate of return, and *Q* is the project's lifetime. C_{inv} , $C_{O&M}$, C_{rep} , C_{EG} , and C_{sv} are the investment, maintenance, replacement, grid, and salvage value costs, respectively, which are calculated in (3)–(7). $C_{special}$ encompasses special costs such as component degradation, transport in difficult-to-access areas, and other specific factors depending on the case under study.

$$C_{inv} = \sum_{k=1}^{K} C_{inv}^k \cdot N^k + C_{dep}$$
(4)

$$C_{O\&M} = \sum_{k=1}^{K} C_{O\&M}^{k} \cdot N^{k} \cdot \sum_{q=1}^{Q} \left(\frac{1+\varepsilon}{1+d}\right)^{q}$$
(5)

$$C_{rep} = \sum_{k=1}^{K} C_{rep}^{k} \cdot N^{k} \cdot \sum_{r=1}^{NOR} \left(\frac{1+\varepsilon}{1+d}\right)^{r}$$
(6)

$$C_{EG} = \left(C_{sub} - \Delta t \cdot \sum_{t_0}^{t_f} \left(p_{EG}^s(t) \cdot \lambda^s(t) + p_{EG}^{in}(t) \cdot \lambda^{in}\right)\right) \cdot \sum_{q=1}^{Q} \left(\frac{1+\varepsilon}{1+d}\right)^q \tag{7}$$

Here, N^k is the total number of the *k*-th microgrid component, and *k* is the index representing the *k*-th microgrid component. C_{inv}^k and C_{dep} are the *k*-th component investment and the deployment cost, respectively. The microgrid deployment cost includes installing the microgrid components such as wiring, concrete, steel, wood, and electrical connections, along with labor and indirect costs associated with the microgrid installation. $C_{O\&M}^k$ is the operation and maintenance cos for the *k*-th component; ε is the escalation rate; and C_{rep}^k is the replacement cost. *NOR* is the number of *r*-th component replacements over the project lifetime, and *q* and *r* are the year and replacement indices, respectively. λ^s and λ^{in} are the grid tariffs for power supply when buying from the grid and injection selling to the grid, respectively. Finally, C_{sub} represents a fixed subscription cost for the power grid, a fixed annual fee imposed by electrical energy provider companies [42]. The salvage cost C_{sv} is considered a cost in favor of the installed capacity of renewable energy. It is generally determined as a percentage of the acquisition cost [41].

3. Microgrid Sizing Approaches

The sizing of microgrids is a complex optimization problem that is typically addressed through a variety of methodologies, as illustrated in Figure 5. These methodologies include commercial software such as HOMER and DER-CAM, heuristic algorithms such as genetic algorithms (GA), PSO, mathematical programming techniques such as mixed integer linear program (MILP), quadratic programming (QP), and hybrid approaches combining the strengths of different methods such as JAYA and GWO, and Simulated Annealing and Tabu

Search. These methodologies are selected for their efficiency in exploring and exploiting the solution space. They balance the trade-offs between computational time, accuracy, and the challenges of handling the multi-dimensional and multi-objective aspects of microgrid sizing. The selection of a specific approach over others depends on the requirements of the microgrid project, including but not limited to cost minimization, reliability maximization, and environmental impact reduction. The advantages and disadvantages of each approach are presented in Table 3, and the following sections address them in detail.



Figure 5. Microgrid sizing approaches.

Table 3. Advantages and disadvantages of each microgrid sizing approach.

Approach	Advantages	Disadvantages
Commercial soft- ware (HOMER)	User-friendly graphical interface, updated to the latest technologies and improvements.	Lacks support for multi-objective problems, does not accommo- date intra-hour variability. Additionally, it exhibits extended com- putational times for intricate problem scenarios.
Heuristic	Heuristic algorithms efficiently address complex opti- mization problems, including sizing problems within a reduced computation time. Moreover, they can handle non-linear equations.	It does not guarantee a global optimum and may get stuck in a local optimum. Additionally, each algorithm has specific pa- rameters that require expert knowledge to be adjusted properly, preventing high computation time and infeasible solutions.
Mathematical	An effective technique to ensure a global optimal solu- tion in the search space.	It cannot solve complex optimization problems due to the substan- tial computation time, thereby limiting server capabilities. Ad- ditionally, it has limitations in handling stochastic environments and is unsuitable for highly complex and non-linear problems.
Hybrid	It converges faster than heuristic methods and takes ad- vantage of each algorithm used. This approach typically addresses multi-dimensional optimization problems	Coordinating the tuning of parameters for each heuristic algo- rithm poses a challenge. There is also a risk of becoming trapped in a local optimum.

3.1. Commercial Software

Commercial software tools play a crucial role in the optimal sizing of microgrids, with the Optimization Model for Electric Renewables (HOMER) standing out as a particularly prominent example. Developed by the National Renewable Energy Laboratory in the United States, HOMER is distinguished by its comprehensive economic evaluation capabilities. Alongside HOMER, there are several other notable tools, including the Hybrid Power System Simulation Model (HYBRID2) by the Renewable Energy Research Laboratory, the General Algebraic Modeling System (GAMS), and the Optimization of Renewable Intermittent Energies with Hydrogen for Autonomous Electrification (ORIENTE), among others. These tools offer diverse approaches and functionalities for sizing microgrids, highlighting their applicability as rich resources for researchers and practitioners.

HOMER software is essential in designing and economically evaluating microgrids, whether or not they are connected to the power grid. It operates through a three-phase simulation, optimization, and sensitivity analysis process. This process utilizes weather data, demand forecasting, and economic and technical considerations, among other inputs, to help identify the optimal sizes of DER for achieving the minimum overall net present value (NPV). This optimization strategy is crucial for designing and evaluating sustainable and efficient energy systems. A notable application of HOMER is in optimizing PV-fuel cells (FCs) microgrids, as discussed in [43]. The study aimed to enhance system efficiency and sustainability through the optimal integration of DER, FCs, and ESSs. The results demonstrated the potential of the proposed microgrid configuration to reduce reliance on conventional energy sources. This configuration also showcased significant improvements in energy management and a reduction in the overall system cost. Similarly, the study in [5] employed HOMER to optimize a microgrid configuration consisting of PV panels, WT, and battery energy storage system (BESS). The optimization focused on minimizing the microgrid's NPV. The system's reliability was also assessed through a sensitivity analysis on the LPSP for evaluating the impact of variations in the microgrid's size and cost.

Research in [44] explored integrating and optimizing multi-microgrids utilizing HOMER Grid software in Goma, DRC. Considering technical and economic aspects, the objective was to minimize the NPV over a 12.5-year horizon. The study highlighted the feasibility of a system configuration without PV, which achieved nearly the same NPV with a significantly lower initial capital cost, albeit with higher operational and maintenance expenses affecting the levelized cost of energy (LCOE). Furthermore, the study in [45] utilized HOMER to determine the optimal configurations for PV/Diesel/Pumphydro and PV/Diesel/BESS systems. This study emphasized calculating payback periods while identifying cost-effective solutions for energy systems. The study in [46] developed a techno-economic methodology for the development of a standalone, renewable energy-based electric vehicle charging station (EVCS) in Qatar by employing HOMER for optimization. The aim was to minimize installation and lifetime operating costs among all technically feasible configurations that met the daily EVCS demand. The optimal configuration's NPV was found to range between \$2.53 M to \$2.92 M, with electricity costs varying from \$0.285 to \$0.329 per kWh. In Vietnam, the optimization of EVCS in major cities was the focus of [47], using HOMER Grid for analysis. The study aimed to reduce NPV and enhance efficiency by considering local solar conditions and economic factors. It proposed optimal system configurations for Hanoi, Da Nang, and Ho Chi Minh City, highlighting the impact of solar irradiation on investment and operational performance.

Additionally, the study in [48] optimized an isolated microgrid for rural electrification in India using GA and HOMER. The study aimed to minimize the NPV and the LCOE by comparing various hybrid configurations. The GA optimization yielded an optimal solution with the lowest LCOE at \$0.163 per kWh. The potential of DER in off-grid microgrid optimization was explored in [49] using MILP and HOMER. The study aimed to minimize the NPV while highlighting significant cost savings and efficiency improvements from implementing DER. Lastly, HOMER Pro, an advanced version of HOMER [50], was used in [51] for optimizing hybrid microgrids for green hydrogen production in Fiji. The study focused on achieving the most cost-effective configuration for hydrogen production, critical for powering fuel cell buses, aiming to minimize the NPV. The economic viability of microgrids across various geographic locations and with different DER combinations has been assessed using HOMER software. Notably, analyses have been conducted on a PV-WT-hydro-biodiesel hybrid system in India [52], a PV-WT-diesel system in Sri Lanka [53], and a WT-biogas hybrid system in Canada [54].

These studies highlight the adaptability and effectiveness of HOMER in optimizing microgrid configurations to improve sustainability, cost-efficiency, and energy reliability. The advantages of HOMER software include:

- User-friendly interface;
- Comprehensive financial analysis capabilities, including cash flow analysis and payback period calculations;
- Integration capability with Geographic Information Systems for enhanced project planning and site selection;
- Advanced sensitivity analysis to evaluate the impact of variable parameters on project outcomes;
- The ability to process hourly data for detailed and accurate modeling of energy systems.
 However, HOMER also has certain limitations:
- Utilizes a "black box" approach to its code, which can limit transparency and customizability;
- Lacks the functionality to formulate multi-objective optimization problems, which limits its applicability in scenarios requiring the balancing of multiple goals;
- Does not account for intra-hour variability, which could affect the accuracy of simulations for systems sensitive to short-term fluctuations;
- It omits the consideration of the DoD BESS, potentially overlooking an important factor in the lifespan and efficiency of ESS, suggesting a need for a more customized and detailed battery degradation model.

3.2. Heuristic Approaches

In the domain of heuristic optimization for microgrid configurations, several studies have employed a variety of algorithms to optimize their design and operational efficiencies. For instance, the work in [41] demonstrated the use of GA to size both grid-connected and isolated microgrids, aiming to optimize the LCOE with configurations including PV panels and BESS. In a similar study presented in [55], GA was applied to optimize a microgrid comprising PV, WT, BESS, and loads. The primary objectives of this optimization were to minimize greenhouse gas emissions, reduce the life cycle cost (LCC), and decrease non-renewable energy consumption. Further research [56] explored multiple microgrid configurations to find a balance between factors such as greenhouse gas emissions versus global cost and microgrid autonomy versus global cost. The study in [57] targeted the sizing of an autonomous AC microgrid to minimize energy and installation costs while enhancing reliability by reducing the LPSP. In addition, ref. [58] aimed at reducing energy and installation costs and maximizing reliability through GA for determining the capacity of BESS, PV, and other components. Lastly, the research presented in [59] discussed integrating life cycle analysis into the hybrid microgrid design process to optimize design and minimize environmental impact while ensuring technical and economic feasibility. The study in [39] critically evaluated a range of metaheuristic algorithms for reducing the NPV of a microgrid system that incorporates residential, commercial, and EV loads, in addition to PV panels, WTs, and BESS. Among these algorithms, the Moth-Flame optimization algorithm stood out for its superior performance, while the Equilibrium optimizer was noted for its relatively lower efficiency. Another study [60] applied the Multi-Objective PSO technique to size the microgrid to minimize the dependency on non-renewable energy imports and the system's annualized cost. The outcomes were illustrated via a Pareto front by highlighting the trade-offs between the two objectives. Furthermore, in [61], a two-step methodology was employed for the optimization and analysis of a standalone hybrid system consisting of PV, WT, BESS, and a diesel generator, which was specifically designed to fulfill the electricity demands of a remote village, Fanisau, in northern Nigeria. This approach, executed in the MATLAB (Version 9.3) environment using a GA solver, aimed to efficiently and economically cater to the unique energy needs of the region. The optimized hybrid renewable energy system proposed for Fanisau included 273 PV modules, 148 batteries, and a 100.31-W diesel generator, capable of producing 200,792 kWh annually at a total annualized cost of \$43,807 USD with a cost of energy \$0.25 USD/kWh. In [62], the study applies the Non-Dominated Sorting GA II (NSGA-II) for microgrid sizing, aiming to minimize annualized costs, emissions, and energy imports from the electrical grid. The

study elucidates optimal configurations via the Pareto front and employs the Topsis method for decision analysis. The study in [63] outlines the development of a hybrid renewable energy system optimization using the PSO algorithm-based Monte Carlo simulation to minimize total annual costs, accounting for resource and load uncertainties. It focuses on microgrid components such as WT, PV panels, and BESS. The findings demonstrate the algorithm's efficiency in enhancing system reliability and cost-effectiveness for offgrid energy solutions. In [64], the grasshopper optimization algorithm (GOA) is used to optimize an autonomous microgrid comprising PV panels, WT, BESS, and a diesel generator. The objective function minimizes the LPSP and the LCOE. The main numerical results demonstrate the GOA efficiency over PSO and Cuckoo Search algorithms. GOA achieves a reduction in system capital cost by 14% and 19.3%, respectively. The optimal configuration includes 26 PV panels, 4 WTs, and a 40 kW BESS. In [65], the article explores microgrid optimization using the fuzzified Grey Wolf optimizer (GWO) to minimize costs and maximize renewable energy use. It covers PV, WT, BESS, and generators, achieving notable cost reductions and efficiency gains, demonstrating significant advancements in sustainable energy management through the application of heuristic algorithms. In the domain of microgrid optimization through metaheuristic techniques, the review paper [66] revealed that approximately 25% of the research in this field employs the PSO method. Additionally, GA and GWO are used by nearly 10% and 5% of the studies analyzed in the paper to enhance the performance of microgrids, respectively.

3.3. Mathematical Approaches

Within the realm of mathematical approaches, MILP has a significant role [67]. This technique is selected for its capacity to secure a globally optimal solution to the optimization problem. In [68], a convex optimization approach was used to determine an island microgrid's optimal sizing and energy management while considering the battery degradation. In [69], the study focuses on a battery's optimal sizing, placement, and daily charge/discharge scheduling within the distribution network. In [70], the paper focuses on utilizing a two-stage stochastic programming. The model addresses uncertainties in DER power and load demand by ensuring a reliable PV power supply for essential services. In [71], an investigation is conducted on a microgrid system in an island territory, which incorporated multiple technologies such as PV, WT, biomass, and geothermal sources, among others, with the objective function of minimizing the overall costs of the system. The authors in [72] conducted a temporal decomposition using Benders' algorithm to determine the optimal sizing and operation of a hybrid railway power substation annually. In [73], the study explores multi-year economic energy planning optimization in microgrids using a MILP method. The authors in [74] utilized linear programming to minimize costs or emissions. They explored the effects of policy on generation investment choices.

As the literature shows, non-linear approaches significantly optimize microgrid systems, where various complex problem formulations are addressed using advanced mathematical models. A mixed integer non-linear program (MINLP) is notably applied to tackle multi-objective optimization problems, such as the one from [75], which focuses on minimizing the system cost related to equipment sizing and emissions. This application highlights the capability of MINLP to simultaneously address multiple aspects of microgrid optimization. It provides solutions that balance economic and environmental concerns. A single objective function is presented in [76] to minimize the system's costs through a non-linear problem formulation. This study extends the application of non-linear optimization to managing multiple microgrids by incorporating probabilistic modeling of energy resources and load demand. Doing so accounts for the inherent uncertainties in energy production and consumption. This approach enhances the decision-making process for microgrid operation and planning. These non-linear approaches underline the complexity and multi-dimensional nature of optimizing microgrid systems. They reflect the ongoing advancements in mathematical modeling techniques to achieve more sustainable and cost-effective energy solutions.

3.4. Hybrid Approaches

Hybrid approaches, which involve the integration of multiple algorithms, are increasingly utilized to improve various aspects of microgrid efficiency. The research presented in [67] employs a synergistic application of the JAYA and GWO algorithms to simultaneously minimize annualized cost and carbon emissions and improve the microgrid's reliability. This approach is recognized for its swift convergence and high accuracy. Conversely, in [77], the authors explore a hybrid model that merges Simulated Annealing and Tabu Search techniques to achieve optimal sizing for autonomous systems, thereby streamlining computational efficiency. Another significant contribution is reported in [78], where a combination of JAYA, PSO, and Harmony Search algorithms is used in designing hybrid DER. This configuration, comprising PV systems, biomass, WT, and ESS, is tailored to meet consumer demand in an efficient, cost-effective, and reliable manner. Additionally, in [79], a hybrid strategy combining NSGA-II and multi-objective PSO is implemented to simultaneously reduce carbon dioxide emissions and the total cost of an isolated microgrid configuration. Furthermore, ref. [80] shows the application of ant colony optimization (ACO) with continuous domain integer programming (CDIP), achieving remarkable accuracy and rapid convergence.

4. Microgrid Energy Management Approaches

The integration of DER into microgrids is beneficial for sustainability, but brings challenges related to reliability and stability because of the unpredictable nature of renewable energy. To mitigate these drawbacks in microgrids, implementing an efficient EMS is essential [81]. Numerous studies have explored energy management within microgrids by employing diverse methodologies. Some of these studies have utilized a rule-based approach, which relies on predefined rules for managing RES and ESS. On the other hand, various studies have adopted optimization-based methods, which seek to find the most efficient solution by optimizing specific objectives, such as minimizing costs or maximizing energy efficiency. The diversity and application of these optimization-based methods within microgrid energy management are illustrated in Figure 6, and the following sections address these in depth.



Figure 6. Microgrid energy management approaches.

4.1. Commercial Software

As with microgrid sizing, commercial software such as HOMER, DER-CAM, and EPLANopt are also used to plan energy management within the microgrid. In [82], the EPLANopt software is utilized alongside a multi-objective evolutionary algorithm to optimize the energy system of Favignana Island for sustainability by the year 2050. EPLANopt models and simulates various energy system configurations, incorporating data such as energy demand and renewable energy potential. The evolutionary algorithm analyzes these configurations, evolving a set of solutions through processes like mutation and selection. While EPLANopt provides detailed simulations, the evolutionary algorithm identifies the most efficient and sustainable solutions by balancing objectives such as cost, emissions, and renewable energy usage. In [83], the RETScreen Expert software is used to examine a grid-connected microgrid wind farm in Ghana, evaluating its technical, financial, and environmental benefits. It investigates the NPV, the internal rate of return, the cost of electricity production, and the emission savings to gauge the project's feasibility and its role in sustainable energy progress. RETScreen Expert software is also used in [84–87]. In [88], a comprehensive study is presented on integrating DER into healthcare systems, focusing on a hospital case study in Azad Jammu and Kashmir. The study uses HOMER Pro and RETScreen Expert software to analyze a proposed hybrid renewable energy system against the current energy setup. The result indicates that HOMER Pro and RETScreen Expert predict the increased energy output with the proposed system, though RETScreen Expert tends to give more optimistic projections. The study highlights the efficiency and sustainability of integrating renewable energy into healthcare while demonstrating significant potential for enhancing infrastructure resilience and reducing costs. HOMER is the most used software in the literature for energy management. In [89], the research details an optimal microgrid design for Basco Island in the Philippines using HOMER software. It integrates PVs, WTs, diesel generators, and BESSs to boost power reliability and sustainability. The study identifies the most efficient and cost-effective system setup by evaluating technical and economic factors. HOMER's simulations, optimizations, and sensitivity analyses are crucial for balancing energy demand, resource availability, and operational cost reduction. In [90], the article explores the best configuration for a hybrid microgrid at a desalination facility in Sfax, Tunisia, by employing HOMER software for simulations and FAHP-PROMETHEE for evaluating multiple criteria. The result shows a dramatic 50.2% cut in energy cost for producing desalinated water, which illustrates hybrid microgrids' significant role in making desalination processes more sustainable and economically viable, especially in areas with inconsistent conventional energy supplies.

4.2. Optimization Based Approach

Figure 6 illustrates the optimization-based strategy for energy management in microgrids. Two categories, metaheuristic and conventional approaches, are defined and analyzed in detail.

4.2.1. Metaheurstic Approaches

Metaheuristic method-based energy management used in microgrid systems is mainly categorized into three major groups: evolutionary algorithms and other metaheuristic algorithms, such as GA and simulated annealing. Evolutionary algorithms, inspired by biological evolution processes such as mutation, recombination, selection, and reproduction, have found extensive applications in computational research. The study in [91] introduces an innovative approach, which proposed a Niching Evolutionary algorithm to optimize the allocation of RES and ESS within isolated microgrids. The research presented in [92] proposed an enhanced memetic algorithm for microgrid demand-side management in microgrids. Another noteworthy contribution is detailed in [93], where the water cycle algorithm is applied to the energy management of microgrids. This approach minimizes operational costs and emissions by efficiently coordinating distributed generation units alongside ESS. Lastly, the study in [94] explores using a GA, the most commonly encoun-

tered evolutionary algorithm in the literature, for optimizing day-ahead demand scheduling under uncertainty. This research highlights the algorithm's robustness in dealing with the stochastic nature of microgrid energy management, further underlining the critical role of evolutionary algorithms in advancing smart city initiatives.

Swarm intelligence draws inspiration from the natural world's social living creatures, which exhibit coordinated behaviors in groups without central control. These phenomena, observable in entities such as ant colonies engaging in foraging, birds moving in unison, animals grouping for migration, hawks coordinating in pursuit of prey, and fish swimming in schools, serve as the foundation for algorithms designed to tackle intricate challenges through collective effort. Among these, the PSO algorithm stands out for its widespread application. This algorithm mimics the social behavior of birds and fish to navigate toward optimal solutions in a multidimensional space, which effectively addresses various computational and optimization problems. The study in [95] presents a probabilistic approach methodology for the enhancement of energy management in microgrids equipped with DER and ESS. The study introduces a self-adaptive modified q-PSO algorithm optimizing the energy management process by adapting to changes dynamically. Thus, it ensures efficient and reliable microgrid operations under diverse conditions. In [96], the study introduces an innovative approach for the energy management of isolated microgrids using a multi-layer ant colony optimization algorithm. This algorithm is designed to minimize electricity production costs by optimizing both day-ahead and real-time scheduling. In [97], the adaptive modified firefly algorithm is introduced for a scenario-based stochastic optimization framework for a microgrid including WT, PV, micro-turbines, FC, and ESS. Beyond swarm optimization and evolutionary algorithm-based metaheuristic techniques, several other metaheuristic algorithms have been employed to enhance the energy management of the microgrid. The study in [98] examines a microgrid in Oshawa, Ontario, Canada, by employing variable load models, which utilize the interior search algorithm to optimize hour-by-hour scheduling for day-ahead power scheduling issues. The twostage scenario-based optimization approach is used for forecasting within a coordinated scheduling model, as detailed in [99].

4.2.2. Conventional Approaches

Conventional approaches have six categories, as illustrated in Figure 6. In [100], a bi-level optimization model is employed where the Karush–Kuhn–Tucker approach for problem reformulation into a single-level optimization is used with a combination of binary PSO and QP. In [101], a dynamic programming solution was implemented to address a convex optimization problem to reduce the microgrid's total operation cost. An energy cost optimization using MILP is introduced in [102]. In the context of model predictive control, a receding horizon control-based model for the optimal scheduling of batteries was introduced in [103]. The study in [104] presents a robust counterpart formulation designed to manage peak demand and smooth out load variations amid energy generation and demand uncertainties within a microgrid using a robust programming strategy. In [105], a stochastic energy scheduling system for a microgrid with DER was assessed, including a case study on a modified IEEE-37 bus test feeder setup. Findings from this research underscored the efficiency and precision of the proposed stochastic programming-based algorithm for microgrid energy scheduling.

4.2.3. Other Optimization Based Approaches

In addition to metaheuristic, conventional, and artificial intelligence (AI) approaches, other methods are also discussed in the literature. Some of these alternative approaches are cited here. In [106], the deployment of an EMS within a standalone microgrid is explored. Utilizing a probabilistic framework grounded in Bayesian networks coupled with Monte Carlo simulations, the study effectively captures the unpredictable aspects of electricity demand, which enhances system profitability. The simulation was conducted for a year, finding that the EMS contributes to a reduction in energy expenses by 11.3% while elevating

the utilization of solar energy to 54%. In [107], the study focuses on enhancing the operation and design of microgrids that incorporate DER using a holistic strategy to minimize energy costs and emissions through the commitment and dispatch of distributed devices. The uncertainties associated with PV generation are captured through a Markovian process, while a branch-and-cut method is applied to address the optimization challenges. In [108], the study presents a methodology employing Hong's Two-Point Estimate method for dayahead scheduling, considering the uncertainties in load consumption and WT generation. The primary goal is to reduce operational costs while maintaining system reliability.

4.3. Other Energy Management Approaches

Rule-based, artificial intelligence, fuzzy logic, game theory, and multi-agent approaches are also utilized for energy management of microgrids. Rule-based approaches, characterized by predefined "if-else" strategies, are extensively explored in the literature for energy management within microgrids. These methodologies often involve establishing a hierarchy among energy sources and storage systems based on user preference or availability. Representative rule-based decision-making in microgrid energy management is presented in Figure 7, widely used in the literature. If the solar and wind output are sufficient, the power is distributed to the load, and any excess power is stored in the BESS. If the renewable energy output is insufficient, power is drawn from the BESS if the battery level is adequate. If the BESS level is low, load shedding is implemented, or power is purchased from the electrical grid. Optionally, the battery discharge can be prioritized to ensure critical loads are powered during nighttime or periods without RES generation. In [56], a priority sequence is determined, allowing PV systems, BESS, or biomass to be selected as the primary energy source contingent upon the user's selection. Similarly, ref. [41] details a rule-based method wherein surplus energy from PV panels is initially stored in the BESS before being dispatched to the electrical grid. They illustrate how rule-based strategies can be adapted to manage diverse configurations of microgrid components, including PV-WT-BESS, PV-Hydrogen, PV-Pumped Hydroelectric, WT-BESS, and PV-BESS systems. Another exemplary implementation of rule-based management strategies is showcased in [109], which proposes a model for an isolated microgrid integrating PV, WT, a diesel generator, and the power grid.

The study in [110] introduces a multi-objective EMS tailored for microgrids, focusing on minimizing overall operational expenses and carbon emissions. This system enhances the BESS management process by utilizing a fuzzy logic approach to manage uncertainties in microgrid operations. It determines the optimal timing and rate for charging or discharging the BESS, considering variables such as RES power production, demand load, and electricity costs. The method employs input membership functions to steer the inference engine in assessing the rules. Furthermore, the study in [111] presents an EMS that utilizes a fuzzy logic controller, which is developed and monitored using LabVIEW, for the regulation of a DC microgrid. This EMS employs fuzzy logic to manage the state of charge (SOC) of lithium-ion batteries, thereby enhancing their life cycle. The study in [112] introduces a strategy for a real-time decentralized demand-side management system in a grid-connected microgrid. Every client linked to the microgrid predicts their daily load demand within this system. Utilizing these predictions, the EMS partakes in a mixed-strategy non-cooperative game, proceeding until a Nash equilibrium is achieved. At this equilibrium, consumption forecasts are modified to aim for the lowest possible electricity cost. In [113], the study proposes a hierarchical control strategy for an isolated microgrid based on a multi-agent system, which focuses on maintaining stable voltage levels while optimizing the economic and environmental aspects of the system. In [114], an EMS model is presented for an isolated hybrid microgrid. The model utilizes PSO in combination with artificial neural networks to enhance energy management strategies. In [115], the authors introduce an EMS powered by recurrent neural networks and complement it with a multi-agent-based weather forecasting method. In [116], a reinforcement learning algorithm was applied to enhance the coordination among various ESS in a microgrid.



Figure 7. Rule-based decision making in microgrid energy management (alt: alternative, opt: optional).

5. Discussion

Optimal microgrid sizing and system energy management can be optimized using a single-stage or a multi-stage methodology. A single-stage optimization approach poses a considerable challenge in promising a globally optimal solution. The wide range of constraints and decision variables that optimization solvers must navigate and the long-term optimization horizon, which could span the project's lifetime, make this approach complex beyond the computational capabilities of conventional computer systems. On the other hand, a multi-stage approach addresses the component sizing dilemma, followed by resolving the energy management challenge in a subsequent phase. This paper discusses single-stage and multi-stage approaches and a multi-objective functions approach in the following sections.

5.1. Single-Stage Co-Optimization of Microgrid Sizing and Energy Management

Figure 8 illustrates the general framework of the single-stage optimization problem. This approach has few references in the literature due to the computational challenges. Within this approach, MILP is one of the most widely used utilities for solving single-stage optimization problems by meeting microgrids' sizing and energy management. In [71], a MILP framework integrates sizing and energy management into one optimization problem by analyzing scenarios for Reunion Island's transition to renewable energy by 2030 and 2050. Another example can be found in [117], where the optimization problem addresses the optimal sizing and energy dispatch of a residential microgrid incorporating PV, a WT, and the power grid connection. In [118], the authors demonstrate a cost optimization approach incorporating hourly dispatch and load demand within an isolated microgrid. In [119], cost minimization is performed through MILP, complemented by an additional sensitivity analysis conducted over the LPSP. Similarly, in [120], a multi-objective function

is formulated using MILP and a trade-off constraint approach, aiming to minimize costs and emissions simultaneously. Concentrating on minimizing the LCOE, the research presented in [121] also formulates the optimization problem using MILP. The results highlight that a microgrid composed of WT, PV, diesel, and BESS exhibits the lowest LCOE compared to alternative combinations. In [122], MILP is similarly employed; the authors explore scenarios wherein a PV-FC-BESS system fulfills residential energy requirements within a grid-connected microgrid environment. The study in [123] introduces a multi-objective optimization approach for the planning and operation of grid-connected microgrids, incorporating DER. It focuses on achieving economic efficiency and enhancing customer satisfaction by implementing demand-side management and employing a fuzzy logic-based method to maximize satisfaction. In [124–126], sizing and energy management are addressed through a single-stage optimization problem using a MILP approach to fully meet the load requirements in grid-connected microgrids and isolated operation modes. The objective function in [124] aims to minimize the cost and emissions over the project lifetime, whereas the objective function in [125,126] focuses solely on cost minimization.



Figure 8. General framework of the single stage problem.

5.2. Multi-Stage Co-Optimization of Microgrid Sizing and Energy Management

In the multi-stage approach, two methodologies are predominantly discussed in the literature. The first methodology begins by addressing the component sizing dilemma, followed by resolving the energy management challenge in a subsequent phase. This sequential methodology leverages optimization-based methods, commercial software solutions, or rule-based strategies for problem formulation. Figure 9 illustrates the outline of this methodology. This methodology has been widely used by researchers. For example, the study [127] contrasts hydrogen and BESS in PV systems by employing a GA for optimal sizing and a rule-based strategy. It focuses on self-sufficiency, NPV, and the impact on the power grid. In [109], the multi-stage methodology is also applied. The stochastic fractal search (SFS) and the symbiotic organisms search (SOS) algorithms are used for sizing, while the energy management uses a rule-based strategy.



Figure 9. General framework of the multi-stage optimization problem.

Another example can be found in [41,55], where GA is used for sizing and a rule-based approach is used for energy management. The NSGA-II with several rule-based options is presented in [56]. The same approach is presented in [60], with the PSO algorithm used for sizing of the grid-connected microgrid. Shifting from heuristic approaches for sizing and rules-based approaches for energy management, some studies have implemented optimization algorithms for both stages. In [100], the model uses the Karush—Kuhn—Tucker approach to simplify the bi-level problem into a single-level one. A mix of binary PSO and QP finds the optimal BESS size and operation schedule to enhance microgrid design and reliability. In [128], the methodology applies the two-stage method, where the first stage focuses on determining the size of the microgrid. In contrast, the second stage addresses the daily energy management considerations. This study implements MILP optimization techniques in both stages. In [129], the first stage employs an evolutionary algorithm to handle the sizing task, exploring a vast solution space to identify viable configurations. In the second stage, MILP fine-tunes these configurations by meticulously scheduling microgrid assets to balance demand fulfillment with the dual objectives of cost minimization and reliability maximization. A similar approach is found in [130], where the study uses a GA to identify optimal sizing solutions that concurrently lower the lifecycle CO_2 emissions and total costs of the microgrid. Moreover, an EMS is developed, employing a MILP algorithm to efficiently distribute power flow while also reducing CO₂ emissions. In [131], a stochastic two-stage model is presented for optimizing microgrid design by addressing the uncertainties in DER. The initial stage focuses on setting capacities for PV, WT, diesel generation, and ESS. In contrast, the second stage targets operational decisions, such as diesel and ESS usage. The study in [132] introduces a method for optimizing microgrid components in "El Espino" community. It uses linear programming (LP), firstly to minimize costs while meeting energy needs, and secondly to incorporate operational and reliability constraints. In [133], the paper applies a two-stage stochastic optimization approach to minimize both capital and operational costs considering uncertainties in demand.

The second form of multi-stage optimization utilizes decomposition algorithms, notably the Dantzig–Wolfe decomposition [134]. This method entails an iterative process that includes several subproblems and a master problem. Each iteration introduces a new variable to the master problem, facilitating a gradual convergence towards the solution. Another key decomposition algorithm is the alternating direction method of multipliers [135], which segments the optimization challenge into multiple sub-problems without incorporating a master problem. Here, each subproblem iteratively shares its solution with the others, aiding in convergence to a global solution. Prominently, Benders' decomposition [136], proposed by Benders [137], is designed to solve mixed-variable optimization problems, with Geoffrion later extending the approach to address non-linear convex problems [138]. Benders' decomposition effectively bifurcates the target problem into two simpler components: the master problem and one or more sub-problems [139]. The master problem, a simplified model, includes a subset of the original variables and constraints, representing a relaxed version of the problem [140]. Conversely, the sub-problems mirror the original problem's complexity, conditioned on the premise that the variables derived from the master problem are fixed.

Figure 10 provides an overview of Benders' decomposition framework. In an optimization problem concerning energy management and component sizing, the master problem involves energy management within the microgrid. At the same time, the sub-problem addresses the sizing of the components. The capacity of each microgrid component is adjusted according to the energy management constraints imposed by the optimization algorithm.

Benders' decomposition algorithm has been used extensively in research. For instance, the study in [141] showcases a multi-stage optimization process for microgrid sizing and energy management using the iterative Benders' decomposition algorithm. In [142], the optimization is decomposed into two principal components: investment decision-making as a master problem and energy management as a sub-problem. By distinguishing between design decision variables and energy management variables, this strategy enhances the

efficiency of solving the planning problem. It iteratively addresses these interconnected components, optimizing microgrid management's investment and operational phases. The study in [143] outlines a two-stage stochastic MILP model for optimizing investments in renewable energy within DER. Its goal is to lower costs associated with investments in PV and WT, as well as operational and total substation expenses, which include energy purchases and losses. This study's innovative aspect is using Benders' decomposition, which simplifies the complex problem by separating investment from operational decisions. In [144], the study presents a parallel multi-period optimal scheduling algorithm for microgrids incorporating ESS and addressing the challenges of DER integration. It optimizes microgrid operations by decomposing inter-temporal constraints and utilizing generalized Benders' and optimality condition decomposition. Based on the optimization models analyzed, the following section delves into the objectives that could be targeted in microgrids' sizing and energy management framework. These objectives cover cost minimization, improved system autonomy, and reduced environmental impact.



Figure 10. General framework of Benders' decomposition algorithm.

5.3. Objective Function

Optimization objectives in energy sizing and management have been addressed in several ways. Some studies focus on minimizing annual costs and cost per kilowatt-hour produced LCOE [145] or NPC [39]. Others aim to improve the autonomy of the microgrid [60] or reduce emissions of CO_2 [59]. Case-specific research considers a single objective. However, some studies address multiple objectives simultaneously. Several approaches can be employed to address these complexities, such as the weighted sum approach, the hierarchical optimization method, the trade-off method, the global criterion method, and the goal programming method. Some approaches for dealing with multiobjective problems simultaneously are described below.

5.3.1. Weighted Sum

- Advantages: The weighted sum approach is appreciated for its simplicity due to its straightforward implementation and understanding. It allows the adjustment of weights according to the significance of each objective, providing a clear way to prioritize among different goals. It simplifies optimization by transforming multiple objectives into a single composite objective using weights. This consolidation renders the multi-objective problem compatible with traditional optimization techniques and tools, which makes the solution process manageable.
- Disadvantages: The effectiveness of the weighted sum approach significantly relies on the selection of weights for each objective. Different sets of weights lead to vastly different solutions; thus, identifying the optimal weights to balance the importance of each objective is critical and challenging. Moreover, when objectives are on substantially different scales, assigning weights that accurately reflect each objective's relative importance becomes complicated. It often necessitates the normalization of objectives to a standard scale, which introduces additional complexity to the optimization process.

5.3.2. Hierarchical Optimization Method

- Advantages: This method's prioritization strategy guarantees that the most critical objectives are addressed first, which is paramount in scenarios where failing to meet primary objectives could negate the relevance of secondary objectives. In environments where some objectives cannot be compromised, such as safety-critical systems, environmental conservation, or healthcare, prioritizing these ensures alignment with the scenario's overarching goals and ethical considerations. This approach ensures that the optimization process respects the most critical requirements.
- Disadvantages: There is a substantial risk that lower-priority objectives might not be considered sufficiently if high-priority objectives significantly consume resources. It could yield solutions that, albeit satisfying primary goals, are sub-optimal in the context of the broader problem. The potential depletion of resources on high-priority objectives might marginalize secondary goals, undermining the comprehensive quality of the solution. Furthermore, optimizing objectives sequentially by importance can introduce challenges when objectives are highly interdependent. Enhancing one objective might impair another. This approach complicates pursuing a balanced and globally optimal solution.

5.3.3. Trade-Off Method

- Advantages: Trade-off analysis aids in identifying Pareto optimal solutions, where improving any objective would lead to the detriment of at least one other objective. It is crucial to ensure that chosen solutions are efficient from a multi-objective standpoint.
- Disadvantages: Analyzing trade-offs can become increasingly complex as the number of objectives grows. This complexity necessitates the use of advanced tools and expertise.

5.3.4. Global Criterion Method

- Advantages: Targets the identification of a solution that optimizes all objectives concurrently, potentially yielding more balanced and universally acceptable outcomes. Moreover, concentrating on a singular, global criterion enhances efficiency in discovering a solution that moderately satisfies all objectives, diminishing the necessity for iterative or repeated optimization processes.
- Disadvantages: Developing a global criterion that accurately encapsulates the significance of all objectives poses a challenge, particularly in scenarios where the objectives significantly conflict or are difficult to quantify. Furthermore, a singular focus on a global criterion may result in missing some Pareto optimal solutions, especially those poorly represented by the selected global criterion.

5.3.5. Goal Programming Method

- Advantages: Goal programming enables concurrently considering various goals, possibly assigning different priorities. This adaptability proves advantageous in intricate decision-making contexts, where navigating trade-offs among conflicting objectives is essential.
- Disadvantages: This approach depends significantly on the decision-maker's capacity to establish and rank goals precisely. Such reliance can inject subjectivity into the process, potentially skewing the results.

A fundamental multi-objective optimization concept is the Pareto front, which represents the trade-offs between conflicting objectives. It draws attention to dominated and non-dominated solutions, demonstrating the trade-offs necessary to achieve optimal results. Each approach to optimizing objectives presents specific advantages and challenges, from the weighted sum to trade-off analysis. The choice of methodology depends on project priorities, so balancing multiple criteria is essential to designing efficient and sustainable microgrids. In addressing uncertainties in renewable generation and demand, advanced optimization techniques such as chance-constrained programming (CCP) have shown potential for real-time adaptation in complex energy systems. Research [146] demonstrates that a CCP-based strategy balances user comfort and system economy in a multi-agent energy system by adjusting confidence levels to manage renewable generation and external temperature uncertainties. This approach could be explored further in microgrid optimization, especially for enhancing cost, reliability, and sustainability trade-offs under uncertainty. The following section presents the key conclusions drawn from this review, highlighting the main contributions and potential avenues for future research.

6. Conclusions

This paper comprehensively reviews the optimal sizing and energy management approaches for energy planning of microgrids. It highlights the importance of detailed energy planning and integrating the sizing and energy management stages to ensure efficient and sustainable operation. The review covers energy planning approaches incorporating renewable energy sources such as PV systems, WTs, and small hydropower plants, and discusses sizing approaches, including heuristic, mathematical, and hybrid methods. Additionally, it emphasizes energy management strategies based on optimization and commercial software. By thoroughly comparing these approaches, this paper underscores the significance of proper sizing in minimizing initial, operational, and maintenance costs, which ensures reliable and stable operation. Hybrid methodologies show high efficiency in solving complex sizing problems and finding optimal solutions. EMS based on algorithms such as genetic algorithms and mathematical programming demonstrate superior efficiency in addressing the power intermittency of renewable sources and enhancing the stability of the energy supply. Commercial software such as HOMER and DER-CAM facilitates energy planning by providing economic evaluation and sensitivity analysis. Furthermore, this review acknowledges the challenges posed by the intermittent power production of

renewable sources, which can lead to instability. However, efficient ESS and EMS mitigate these issues and improve supply reliability.

Two paradigms are identified when looking at multi-stage and single-stage sizing and energy management optimization approaches. The multi-stage approach offers the advantage of modularity and simplification, where component sizing is addressed first, followed by energy management in the second stage. It allows the application of algorithms such as the Dantzig–Wolfe decomposition, which segments the problem into subproblems and facilitates iterative convergence. On the other hand, in the single-stage approach, simultaneous integration of sizing and energy management decisions can lead to optimal solutions, but high computational demands limit the process. This methodology directly addresses the simultaneity of decisions, resulting in a more efficient but computationally expensive overall optimization. In addition, decomposition algorithms, such as Benders', allow complex problems to be broken down into more manageable parts, facilitating the solution of nonlinear and mixed problems. The advantage of this methodology is that it allows for handling uncertainties in renewable energy production and provides a robust approach to energy planning.

This paper contributes to microgrid energy planning and management. First, it provides a comparative analysis of heuristic, mathematical, and hybrid sizing methodologies, underscoring their applicability in different operational contexts. Second, it introduces a detailed discussion of energy management strategies based on optimization techniques, offering practical insights for addressing the challenges associated with integrating renewable energy sources. By combining sizing and energy management stages, this work presents a framework for optimizing cost and performance in microgrid systems. Furthermore, it highlights the potential of multi-objective optimization techniques, which are essential for improving system autonomy, reducing carbon emissions, and ensuring the reliability of microgrids in diverse environments. One of the existing challenges identified in this review is the difficulty of implementing real-time control strategies that effectively leverage optimization results while managing uncertainties. The dynamic nature of microgrid operation, combined with the variability of RES, introduces substantial complexity into real-time decision-making processes. In addition, uncertainties, such as those related to fluctuating energy demand, unpredictable weather conditions, and degradation of the ESS, pose a considerable obstacle to the practical application of optimization techniques. These uncertainties could lead to suboptimal decisions if not adequately considered in the microgrid's planning and operational phases.

This review highlighted the advantage of developing methodologies that integrate the sizing and energy management stages and address them simultaneously, which could provide more accurate solutions. Effective integration of the sizing and energy management stages, with the use of advanced methodologies and emerging technologies, will significantly improve the efficiency and reliability of microgrids. Integrating technologies such as electric vehicles and smart grids into the planning and operation of microgrids is a promising avenue for energy efficiency and sustainability. Future research could focus on improving the robustness of optimization algorithms to handle uncertainties better, ensuring that real-time control systems can respond dynamically to changing conditions without compromising the stability or efficiency of the microgrid. Research in this area could provide opportunities for microgrid planning and energy management optimization. Also, upcoming works could address multi-objective optimization, including cost minimization, CO_2 emission reduction, and autonomy. Advanced multi-objective energy management techniques could significantly improve energy planning.

Author Contributions: Conceptualization, R.R. and B.C.; funding acquisition, F.L. and M.S.; methodology, B.C., F.L. and M.S.; resources, F.L. and M.S.; supervision, B.C., F.L. and M.S.; writing—original draft, F.A.K.; writing—review & editing, R.R. and B.C. All authors have read and agreed to the published version of the manuscript. Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not Applicable.

Conflicts of Interest: The authors declare no conflicts of interest.

References

- 1. Boche, A.; Foucher, C.; Villa, L.F.L. Understanding microgrid sustainability: A systemic and comprehensive review. *Energies* 2022, 15, 2906. [CrossRef]
- Jirdehi, M.A.; Tabar, V.S.; Ghassemzadeh, S.; Tohidi, S. Different aspects of microgrid management: A comprehensive review. J. Energy Storage 2020, 30, 101457. [CrossRef]
- 3. Shahzad, S.; Abbasi, M.A.; Ali, H.; Iqbal, M.; Munir, R.; Kilic, H. Possibilities, challenges, and future opportunities of microgrids: A review. *Sustainability* **2023**, *15*, 6366. [CrossRef]
- 4. Saeed, M.H.; Fangzong, W.; Kalwar, B.A.; Iqbal, S. A review on microgrids' challenges perspectives. *IEEE Access* 2021, 9, 166502–166517. [CrossRef]
- Hosseini, S.J.a.D.; Moazzami, M.; Shahinzadeh, H. Optimal sizing of an isolated hybrid wind/PV/battery system with considering loss of power supply probability. *Majlesi J. Electr. Eng.* 2017, 11, 63–69.
- Budiman, F.N.; Ramli, M.A.; Bouchekara, H.R.; Milyani, A.H. Optimal scheduling of a microgrid with power quality constraints based on demand side management under grid-connected and islanding operations. *Int. J. Electr. Power Energy Syst.* 2024, 155, 109650. [CrossRef]
- Ahmadipour, M.; Murtadha Othman, M.; Salam, Z.; Alrifaey, M.; Mohammed Ridha, H.; Veerasamy, V. Optimal load shedding scheme using grasshopper optimization algorithm for islanded power system with distributed energy resources. *Ain Shams Eng. J.* 2023, 14, 101835. [CrossRef]
- Booth, S.; Reilly, J.; Butt, R.; Wasco, M.; Monohan, R. Microgrids for Energy Resilience: A Guide to Conceptual Design and Lessons from Defense Projects Microgrids for Energy Resilience; Technical Report; National Renewable Energy Laboratory (NREL): Golden, CO, USA, 2019. Available online: https://www.nrel.gov/docs/fy19osti/72586.pdf (accessed on 28 August 2024).
- 9. Kamal, M.M.; Ashraf, I.; Fernandez, E. Optimal sizing of standalone rural microgrid for sustainable electrification with renewable energy resources. *Sustain. Cities Soc.* 2023, *88*, 104298. [CrossRef]
- 10. Premadasa, P.; Silva, C.; Chandima, D.; Karunadasa, J. A multi-objective optimization model for sizing an off-grid hybrid energy microgrid with optimal dispatching of a diesel generator. *J. Energy Storage* **2023**, *68*, 107621. [CrossRef]
- 11. Rodriguez, R.; Osma, G.; Ordoñez, G. Sizing of a scattered housing microgrid in a remote rural area. *Renew. Energy Power Qual. J.* **2022**, *20*, 43–48. [CrossRef]
- 12. Shalom Irence, L.B.; Immanuel Selvakumar, A. Optimal Sizing of Various Types of Microgrids—A Review. *J. Eng. Sci. Technol. Rev.* 2021, *14*, 176–192. [CrossRef]
- 13. Liu, X.; Zhao, T.; Deng, H.; Wang, P.; Liu, J.; Blaabjerg, F. Microgrid Energy Management with Energy Storage Systems: A Review. *CSEE J. Power Energy Syst.* 2023, 9, 483–504. [CrossRef]
- 14. Abdi, H. A Brief Review of Microgrid Surveys, by Focusing on Energy Management System. *Sustainability* **2023**, *15*, 284. [CrossRef]
- 15. Shafiullah, M.; Refat, A.M.; Haque, M.E.; Chowdhury, D.M.H.; Hossain, M.S.; Alharbi, A.G.; Alam, M.S.; Ali, A.; Hossain, S. Review of Recent Developments in Microgrid Energy Management Strategies. *Sustainability* **2022**, *14*, 14794. [CrossRef]
- Ahmad, S.; Shafiullah, M.; Ahmed, C.B.; Alowaifeer, M. A Review of Microgrid Energy Management and Control Strategies. IEEE Access 2023, 11, 21729–21757. [CrossRef]
- 17. *IEEE Std* 1547-2003; IEEE Standard for Interconnecting Distributed Resources with Electric Power Systems. IEEE: New York, NY, USA, 2003.
- 18. *IEEE Std* 1547.3-2007; IEEE Guide for Monitoring, Information Exchange, and Control of Distributed Resources Interconnected with Electric Power Systems. IEEE: New York, NY, USA, 2007.
- 19. *IEEE Std* 1547.6-2011; IEEE Recommended Practice for Interconnecting Distributed Resources with Electric Power Systems Distribution Secondary Networks. IEEE: New York, NY, USA, 2011.
- 20. *IEEE Std* 2030-2011; IEEE Guide for Control and Automation Installations Applied to the Electric Power Infrastructure. IEEE: New York, NY, USA, 2011.
- Battula, A.R.; Vuddanti, S.; Salkuti, S.R. Review of Energy Management System Approaches in Microgrids. *Energies* 2021, 14, 5459. [CrossRef]
- 22. Cabana-Jiménez, K.; Candelo-Becerra, J.E.; Sousa-Santos, V. Comprehensive Analysis of Microgrids Configurations and Topologies. *Sustainability* **2022**, *14*, 1056. [CrossRef]
- 23. Mirzaeva, G.; Miller, D. DC and AC Microgrids for Standalone Applications. *IEEE Trans. Ind. Appl.* **2023**, *59*, 7908–7918. [CrossRef]

- 24. Hmad, J.; Houari, A.; Bouzid, A.E.M.; Saim, A.; Trabelsi, H. A Review on Mode Transition Strategies between Grid-Connected and Standalone Operation of Voltage Source Inverters-Based Microgrids. *Energies* **2023**, *16*, 5062. [CrossRef]
- Panda, S.K.; Subudhi, B. A Review on Robust and Adaptive Control Schemes for Microgrid. J. Mod. Power Syst. Clean Energy 2023, 11, 1027–1040. [CrossRef]
- Polleux, L.; Guerassimoff, G.; Marmorat, J.P.; Sandoval-Moreno, J.; Schuhler, T. An overview of the challenges of solar power integration in isolated industrial microgrids with reliability constraints. *Renew. Sustain. Energy Rev.* 2022, 155, 111955. [CrossRef]
- 27. Uzair, M.; Li, L.; Eskandari, M.; Hossain, J.; Zhu, J.G. Challenges, advances and future trends in AC microgrid protection: With a focus on intelligent learning methods. *Renew. Sustain. Energy Rev.* **2023**, *178*, 113228. [CrossRef]
- Jithin, K.; Haridev, P.P.; Mayadevi, N.; Harikumar, R.P.; Mini, V.P. A Review on Challenges in DC Microgrid Planning and Implementation. J. Mod. Power Syst. Clean Energy 2023, 11, 1375–1395. [CrossRef]
- 29. Charadi, S.; Chaibi, Y.; Redouane, A.; Allouhi, A.; El Hasnaoui, A.; Mahmoudi, H. Efficiency and energy-loss analysis for hybrid AC/DC distribution systems and microgrids: A review. *Int. Trans. Electr. Energy Syst.* **2021**, *31*, e13203. [CrossRef]
- Renewable Energy Policy Network for the 21st Century. Renewable Energy and Sustainability Report Renewable Energy and Sustainability Report. Technical Report. REN21. 2024. Available online: https://www.ren21.net/renewable-energy-and-sustainability-report-2024 (accessed on 28 August 2024).
- International Energy Agency. *Renewables* 2023; Technical Report; IEA: Paris, France, 2024. Available online: https://www.iea. org/reports/renewables-2023 (accessed on 28 August 2024).
- 32. International Renewable Energy Agency. *Renewable Energy Statistics* 2024; Technical Report; IRENA: Abu Dhabi, United Arab Emirates, 2024. Available online: http://www.irena.org/Publications/2024/Jul/Renewable-energy-statistics-2024 (accessed on 28 August 2024).
- Vivek, C.M.; Ramkumar, P.; Srividhya, P.K.; Sivasubramanian, M. Recent strategies and trends in implanting of renewable energy sources for sustainability—A review. *Mater. Today Proc.* 2021, 46, 8204–8208. [CrossRef]
- 34. Alam, M.S.; Hossain, M.A.; Shafiullah, M.; Islam, A.; Choudhury, M.S.; Faruque, M.O.; Abido, M.A. Renewable energy integration with DC microgrids: Challenges and opportunities. *Electr. Power Syst. Res.* **2024**, 234, 110548. [CrossRef]
- Shahgholian, G. A brief review on microgrids: Operation, applications, modeling, and control. *Int. Trans. Electr. Energy Syst.* 2021, 31, e12885. [CrossRef]
- Gutiérrez-Oliva, D.; Colmenar-Santos, A.; Rosales-Asensio, E. A Review of the State of the Art of Industrial Microgrids Based on Renewable Energy. *Electronics* 2022, 11, 1002. [CrossRef]
- 37. Frydrychowicz-Jastrzebska, G. El Hierro renewable energy hybrid system: A tough compromise. *Energies* **2018**, *11*, 2812. [CrossRef]
- de la Cruz, J.; Wu, Y.; Candelo-Becerra, J.E.; Vasquez, J.C.; Guerrero, J.M. Review of Networked Microgrid Protection: Architectures, Challenges, Solutions, and Future Trends. CSEE J. Power Energy Syst. 2024, 10, 448–467. [CrossRef]
- Mohseni, S.; Khalid, R.; Brent, A.C. Stochastic, resilience-oriented optimal sizing of off-grid microgrids considering EV-charging demand response: An efficiency comparison of state-of-the-art metaheuristics. *Appl. Energy* 2023, 341, 121007. [CrossRef]
- 40. Saha, D.; Bazmohammadi, N.; Raya-Armenta, J.M.; Bintoudi, A.D.; Lashab, A.; Vasquez, J.C.; Guerrero, J.M. Optimal Sizing and Siting of PV and Battery Based Space Microgrids Near the Moon's Shackleton Crater. *IEEE Access* **2023**, *11*, 8701–8717. [CrossRef]
- 41. Hassan, A.; Al-Abdeli, Y.M.; Masek, M.; Bass, O. Optimal sizing and energy scheduling of grid-supplemented solar PV systems with battery storage: Sensitivity of reliability and financial constraints. *Energy* **2022**, 238, 121780. [CrossRef]
- Bagheri, F.; Dagdougui, H.; Gendreau, M. Stochastic optimization and scenario generation for peak load shaving in Smart District microgrid: Sizing and operation. *Energy Build.* 2022, 275, 112426. [CrossRef]
- Jamalaiah, A.; Raju, C.P.; Srinivasarao, R. Optimization and operation of a renewable energy based pv-fc-micro grid using homer. In Proceedings of the 2017 International Conference on Inventive Communication and Computational Technologies (ICICCT), Coimbatore, India, 10–11 March 2017; pp. 450–455. [CrossRef]
- 44. Khan Ngwashi, D.; Kafatiya Arnold, A.; Godwill Ndeh, S.; Tanyi, E. Optimal design and sizing of a multi-microgrids system: Case study of Goma in The Democratic Republic of the Congo. *Sci. Afr.* **2023**, *22*, e01913. [CrossRef]
- 45. Das, B.K.; Hasan, M.; Rashid, F. Optimal sizing of a grid-independent PV/diesel/pump-hydro hybrid system: A case study in Bangladesh. *Sustain. Energy Technol. Assess.* **2021**, *44*, 100997. [CrossRef]
- 46. Al Wahedi, A.; Bicer, Y. Techno-economic optimization of novel stand-alone renewables-based electric vehicle charging stations in Qatar. *Energy* **2022**, *243*, 123008. [CrossRef]
- 47. Minh, P.V.; Le Quang, S.; Pham, M.H. Technical Economic Analysis of Photovoltaic-Powered Electric Vehicle Charging Stations under Different Solar Irradiation Conditions in Vietnam. *Sustainability* **2021**, *13*, 3528. [CrossRef]
- 48. Suresh, V.; Muralidhar, M.; Kiranmayi, R. Modelling and optimization of an off-grid hybrid renewable energy system for electrification in a rural areas. *Energy Rep.* 2020, *6*, 594–604. [CrossRef]
- 49. Amrollahi, M.H.; Bathaee, S.M.T. Techno-economic optimization of hybrid photovoltaic/wind generation together with energy storage system in a stand-alone micro-grid subjected to demand response. *Appl. Energy* **2017**, 202, 66–77. [CrossRef]
- 50. HOMER. HOMER Software. 2024. Available online: https://homerenergy.com (accessed on 28 August 2024).
- 51. Ram, K.; Chand, S.S.; Prasad, R.; Mohammadi, A.; Cirrincione, M. Microgrids for green hydrogen production for fuel cell buses—A techno-economic analysis for Fiji. *Energy Convers. Manag.* **2024**, *300*, 117928. [CrossRef]

- 52. Sen, R.; Bhattacharyya, S.C. Off-grid electricity generation with renewable energy technologies in India: An application of HOMER. *Renew. Energy* 2014, *62*, 388–398. [CrossRef]
- 53. Kolhe, M.L.; Ranaweera, K.I.U.; Gunawardana, A.S. Techno-economic sizing of off-grid hybrid renewable energy system for rural electrification in Sri Lanka. *Sustain. Energy Technol. Assess.* **2015**, *11*, 53–64. [CrossRef]
- Mudasser, M.; Yiridoe, E.K.; Corscadden, K. Cost-benefit analysis of grid-connected wind-biogas hybrid energy production, by turbine capacity and site. *Renew. Energy* 2015, 80, 573–582. [CrossRef]
- 55. Khlifi, F.; Cherif, H.; Belhadj, J. Environmental and Economic Optimization and Sizing of a Micro-Grid with Battery Storage for an Industrial Application. *Energies* 2021, 14, 5913. [CrossRef]
- Dougier, N.; Garambois, P.; Gomand, J.; Roucoules, L. Multi-objective non-weighted optimization to explore new efficient design of electrical microgrids. *Appl. Energy* 2021, 304, 117758. [CrossRef]
- 57. Çetinbaş, İ.; Tamyürek, B.; Demirtaş, M. Sizing optimization and design of an autonomous AC microgrid for commercial loads using Harris Hawks Optimization algorithm. *Energy Convers. Manag.* **2021**, 245, 114562. [CrossRef]
- Alramlawi, M.; Li, P. Design Optimization of a Residential PV-Battery Microgrid with a Detailed Battery Lifetime Estimation Model. *IEEE Trans. Ind. Appl.* 2020, 56, 2020–2030. [CrossRef]
- Jiménez-Vargas, I.; Rey, J.M.; Osma-Pinto, G. Sizing of hybrid microgrids considering life cycle assessment. *Renew. Energy* 2023, 202, 554–565. [CrossRef]
- 60. Duchaud, J.L.; Notton, G.; Darras, C.; Voyant, C. Multi-Objective Particle Swarm optimal sizing of a renewable hybrid power plant with storage. *Renew. Energy* **2019**, *131*, 1156–1167. [CrossRef]
- Yimen, N.; Tchotang, T.; Kanmogne, A.; Abdelkhalikh Idriss, I.; Musa, B.; Aliyu, A.; Okonkwo, E.C.; Abba, S.I.; Tata, D.; Meva'a, L.; et al. Optimal Sizing and Techno-Economic Analysis of Hybrid Renewable Energy Systems—A Case Study of a Photovoltaic/Wind/Battery/Diesel System in Fanisau, Northern Nigeria. *Processes* 2020, *8*, 1381. [CrossRef]
- 62. Wang, Y.; Wang, X.; Yu, H.; Huang, Y.; Dong, H.; Qi, C.; Baptiste, N. Optimal design of integrated energy system considering economics, autonomy and carbon emissions. *J. Clean. Prod.* **2019**, *225*, 563–578. [CrossRef]
- 63. Maleki, A.; Khajeh, M.G.; Ameri, M. Optimal sizing of a grid independent hybrid renewable energy system incorporating resource uncertainty, and load uncertainty. *Int. J. Electr. Power Energy Syst.* 2016, 83, 514–524. [CrossRef]
- Bukar, A.L.; Tan, C.W.; Lau, K.Y. Optimal sizing of an autonomous photovoltaic/wind/battery/diesel generator microgrid using grasshopper optimization algorithm. *Sol. Energy* 2019, 188, 685–696. [CrossRef]
- 65. Mukhopadhyay, B.; Das, D. Optimal multi-objective long-term sizing of distributed energy resources and hourly power scheduling in a grid-tied microgrid. *Sustain. Energy Grids Netw.* **2022**, *30*, 100632. [CrossRef]
- Akter, A.; Zafir, E.I.; Dana, N.H.; Joysoyal, R.; Sarker, S.K.; Li, L.; Muyeen, S.M.; Das, S.K.; Kamwa, I. A review on microgrid optimization with meta-heuristic techniques: Scopes, trends and recommendation. *Energy Strategy Rev.* 2024, 51, 101298. [CrossRef]
- 67. Alzahrani, A.; Hayat, M.A.; Khan, A.; Hafeez, G.; Khan, F.A.; Khan, M.I.; Ali, S. Optimum sizing of stand-alone microgrids: Wind turbine, solar photovoltaic, and energy storage system. *J. Energy Storage* **2023**, *73*, 108611. [CrossRef]
- Huo, D.; Santos, M.; Sarantakos, I.; Resch, M.; Wade, N.; Greenwood, D. A reliability-aware chance-constrained battery sizing method for island microgrid. *Energy* 2022, 251, 123978. [CrossRef]
- 69. Jannesar, M.R.; Sedighi, A.; Savaghebi, M.; Guerrero, J.M. Optimal placement, sizing, and daily charge/discharge of battery energy storage in low voltage distribution network with high photovoltaic penetration. *Appl. Energy* **2018**, 226, 957–966. [CrossRef]
- 70. Kizito, R.; Liu, Z.; Li, X.; Sun, K. Stochastic optimization of distributed generator location and sizing in an islanded utility microgrid during a large-scale grid disturbance. *Sustain. Energy Grids Netw.* **2021**, *27*, 100516. [CrossRef]
- 71. François, A.; Roche, R.; Grondin, D.; Benne, M. Assessment of medium and long term scenarios for the electrical autonomy in island territories: The Reunion Island case study. *Renew. Energy* **2023**, *216*, 119093. [CrossRef]
- 72. Brisset, S.; Ogier, M. Collaborative and multilevel optimizations of a hybrid railway power substation. *Int. J. Numer. Model. Electron. Netw. Devices Fields* **2019**, *32*, e2289. [CrossRef]
- 73. Pecenak, Z.K.; Stadler, M.; Fahy, K. Efficient multi-year economic energy planning in microgrids. *Appl. Energy* **2019**, 255, 113771. [CrossRef]
- 74. Guo, Z.; Cheng, R.; Xu, Z.; Liu, P.; Wang, Z.; Li, Z.; Jones, I.; Sun, Y. A multi-region load dispatch model for the long-term optimum planning of China's electricity sector. *Appl. Energy* **2017**, *185*, 556–572. [CrossRef]
- 75. Rezvani, A.; Gandomkar, M.; Izadbakhsh, M.; Ahmadi, A. Environmental/economic scheduling of a micro-grid with renewable energy resources. *J. Clean. Prod.* 2015, *87*, 216–226. [CrossRef]
- Nikmehr, N.; Najafi Ravadanegh, S. Optimal Power Dispatch of Multi-Microgrids at Future Smart Distribution Grids. *IEEE Trans.* Smart Grid 2015, 6, 1648–1657. [CrossRef]
- Katsigiannis, Y.A.; Georgilakis, P.S.; Karapidakis, E.S. Hybrid Simulated Annealing–Tabu Search Method for Optimal Sizing of Autonomous Power Systems with Renewables. *IEEE Trans. Sustain. Energy* 2012, *3*, 330–338. [CrossRef]
- 78. Alshammari, N.; Asumadu, J. Optimum unit sizing of hybrid renewable energy system utilizing harmony search, Jaya and particle swarm optimization algorithms. *Sustain. Cities Soc.* **2020**, *60*, 102255. [CrossRef]
- 79. Lan, H.; Wen, S.; Hong, Y.Y.; Yu, D.C.; Zhang, L. Optimal sizing of hybrid PV/diesel/battery in ship power system. *Appl. Energy* **2015**, *158*, 26–34. [CrossRef]

- 80. Fetanat, A.; Khorasaninejad, E. Size optimization for hybrid photovoltaic–wind energy system using ant colony optimization for continuous domains based integer programming. *Appl. Soft Comput.* **2015**, *31*, 196–209. [CrossRef]
- Thirunavukkarasu, G.S.; Seyedmahmoudian, M.; Jamei, E.; Horan, B.; Mekhilef, S.; Stojcevski, A. Role of optimization techniques in microgrid energy management systems—A review. *Energy Strategy Rev.* 2022, 43, 100899. [CrossRef]
- Groppi, D.; Nastasi, B.; Prina, M.G.; Astiaso Garcia, D. The EPLANopt model for Favignana island's energy transition. *Energy Convers. Manag.* 2021, 241, 114295. [CrossRef]
- 83. Asamoah, S.S.; Parbey, J.; Yankey, I.K.; Awuah, A. Techno-economic assessment of a central grid-connected wind farm in Ghana using RETScreen[®] Expert. *Heliyon* 2023, *9*, e12902. [CrossRef]
- Owolabi, A.B.; Emmanuel Kigha Nsafon, B.; Wook Roh, J.; Suh, D.; Huh, J.S. Measurement and verification analysis on the energy performance of a retrofit residential building after energy efficiency measures using RETScreen Expert. *Alex. Eng. J.* 2020, 59, 4643–4657. [CrossRef]
- Owolabi, A.B.; Nsafon, B.E.K.; Roh, J.W.; Suh, D.; Huh, J.S. Validating the techno-economic and environmental sustainability of solar PV technology in Nigeria using RETScreen Experts to assess its viability. *Sustain. Energy Technol. Assess.* 2019, 36, 100542. [CrossRef]
- Zandi, M.; Bahrami, M.; Eslami, S.; Gavagsaz-Ghoachani, R.; Payman, A.; Phattanasak, M.; Nahid-Mobarakeh, B.; Pierfederici, S. Evaluation and comparison of economic policies to increase distributed generation capacity in the Iranian household consumption sector using photovoltaic systems and RETScreen software. *Renew. Energy* 2017, 107, 215–222. [CrossRef]
- Alhassan, M.O.; Opoku, R.; Uba, F.; Obeng, G.Y.; Sekyere, C.K.; Nyanor, P. Techno-economic and environmental estimation assessment of floating solar PV power generation on Akosombo dam reservoir in Ghana. *Energy Rep.* 2023, 10, 2740–2755. [CrossRef]
- 88. Naveed, A.; Iqbal, S.; Munir, S.; Rehman, A.u.; Eslami, M.; Kamel, S. Renewable energy integration in healthcare systems: A case study of a hospital in Azad Jammu and Kashmir. *IET Renew. Power Gener.* **2024**, *18*, 796–809. [CrossRef]
- 89. Lin, C.E.; Phan, B.C.; Lai, Y.C. Optimal Design of Hybrid Renewable Energy System Using HOMER: A Case Study in the Philippines. In Proceedings of the 2019 SoutheastCon, Huntsville, AL, USA, 11–14 April 2019; pp. 1–6.
- Mallek, M.; Elleuch, M.A.; Euchi, J.; Jerbi, Y. Optimum design of on-grid PV/wind hybrid system for desalination plant: A case study in Sfax, Tunisia. *Desalination* 2024, 576, 117358. [CrossRef]
- Conti, S.; Nicolosi, R.; Rizzo, S.A.; Zeineldin, H.H. Optimal Dispatching of Distributed Generators and Storage Systems for MV Islanded Microgrids. *IEEE Trans. Power Deliv.* 2012, 27, 1243–1251. [CrossRef]
- 92. Li, P.; Zheng, M. Multi-objective optimal operation of hybrid AC/DC microgrid considering source-network-load coordination. *J. Mod. Power Syst. Clean Energy* **2019**, *7*, 1229–1240. [CrossRef]
- 93. Deihimi, A.; Keshavarz Zahed, B.; Iravani, R. An interactive operation management of a micro-grid with multiple distributed generations using multi-objective uniform water cycle algorithm. *Energy* **2016**, *106*, 482–509. [CrossRef]
- 94. Battula, A.R.; Vuddanti, S.; Salkuti, S.R. A Day Ahead Demand Schedule Strategy for Optimal Operation of Microgrid with Uncertainty. *Smart Cities* 2023, *6*, 491–509. [CrossRef]
- 95. Baziar, A.; Kavousi-Fard, A. Considering uncertainty in the optimal energy management of renewable micro-grids including storage devices. *Renew. Energy* 2013, 59, 158–166. [CrossRef]
- Marzband, M.; Yousefnejad, E.; Sumper, A.; Domínguez-García, J.L. Real time experimental implementation of optimum energy management system in standalone Microgrid by using multi-layer ant colony optimization. *Int. J. Electr. Power Energy Syst.* 2016, 75, 265–274. [CrossRef]
- 97. Mohammadi, S.; Soleymani, S.; Mozafari, B. Scenario-based stochastic operation management of MicroGrid including Wind, Photovoltaic, Micro-Turbine, Fuel Cell and Energy Storage Devices. *Int. J. Electr. Power Energy Syst.* 2014, 54, 525–535. [CrossRef]
- 98. El-Hendawi, M.; Gabbar, H.A.; El-Saady, G.; Ibrahim, E.N.A. Control and EMS of a Grid-Connected Microgrid with Economical Analysis. *Energies* **2018**, *11*, 129. [CrossRef]
- Jiang, Y.; Wan, C.; Chen, C.; Shahidehpour, M.; Song, Y. A Hybrid Stochastic-Interval Operation Strategy for Multi-Energy Microgrids. *IEEE Trans. Smart Grid* 2020, 11, 440–456. [CrossRef]
- 100. Hayashi, R.; Takano, H.; Nyabuto, W.M.; Asano, H.; Nguyen-Duc, T. Bilevel optimization model for sizing of battery energy storage systems in a microgrid considering their economical operation. *Energy Rep.* **2023**, *9*, 728–737. [CrossRef]
- 101. Rahbar, K.; Xu, J.; Zhang, R. Real-Time Energy Storage Management for Renewable Integration in Microgrid: An Off-Line Optimization Approach. *IEEE Trans. Smart Grid* 2015, *6*, 124–134. [CrossRef]
- 102. Cheikh-Mohamad, S.; Sechilariu, M.; Locment, F. Real-Time Power Management Including an Optimization Problem for PV-Powered Electric Vehicle Charging Stations. *Appl. Sci.* **2022**, *12*, 4323. [CrossRef]
- 103. Wu, Z.; Tazvinga, H.; Xia, X. Demand side management of photovoltaic-battery hybrid system. *Appl. Energy* **2015**, *148*, 294–304. [CrossRef]
- Malysz, P.; Sirouspour, S.; Emadi, A. An Optimal Energy Storage Control Strategy for Grid-connected Microgrids. *IEEE Trans.* Smart Grid 2014, 5, 1785–1796. [CrossRef]
- Su, W.; Wang, J.; Roh, J. Stochastic Energy Scheduling in Microgrids with Intermittent Renewable Energy Resources. *IEEE Trans.* Smart Grid 2014, 5, 1876–1883. [CrossRef]

- 106. Thiaux, Y.; Dang, T.T.; Schmerber, L.; Multon, B.; Ben Ahmed, H.; Bacha, S.; Tran, Q.T. Demand-side management strategy in stand-alone hybrid photovoltaic systems with real-time simulation of stochastic electricity consumption behavior. *Appl. Energy* 2019, 253, 113530. [CrossRef]
- 107. Yan, B.; Luh, P.B.; Warner, G.; Zhang, P. Operation and Design Optimization of Microgrids with Renewables. *IEEE Trans. Autom. Sci. Eng.* 2017, 14, 573–585. [CrossRef]
- 108. Ghahramani, M.; Nazari-Heris, M.; Zare, K.; Mohammadi-Ivatloo, B. Energy and reserve management of a smart distribution system by incorporating responsive-loads/battery/wind turbines considering uncertain parameters. *Energy* 2019, 183, 205–219. [CrossRef]
- Bacha, B.; Ghodbane, H.; Dahmani, H.; Betka, A.; Toumi, A.; Chouder, A. Optimal sizing of a hybrid microgrid system using solar, wind, diesel, and battery energy storage to alleviate energy poverty in a rural area of Biskra, Algeria. *J. Energy Storage* 2024, 84, 110651. [CrossRef]
- 110. Chaouachi, A.; Kamel, R.M.; Andoulsi, R.; Nagasaka, K. Multiobjective Intelligent Energy Management for a Microgrid. *IEEE Trans. Ind. Electron.* **2013**, *60*, 1688–1699. [CrossRef]
- Chen, Y.K.; Wu, Y.C.; Song, C.C.; Chen, Y.S. Design and Implementation of Energy Management System with Fuzzy Control for DC Microgrid Systems. *IEEE Trans. Power Electron.* 2013, 28, 1563–1570. [CrossRef]
- 112. Tushar, M.H.K.; Zeineddine, A.W.; Assi, C. Demand-Side Management by Regulating Charging and Discharging of the EV, ESS, and Utilizing Renewable Energy. *IEEE Trans. Ind. Inform.* **2018**, *14*, 117–126. [CrossRef]
- 113. Dou, C.X.; Liu, B. Multi-Agent Based Hierarchical Hybrid Control for Smart Microgrid. *IEEE Trans. Smart Grid* 2013, 4, 771–778. [CrossRef]
- Aguila-Leon, J.; Vargas-Salgado, C.; Chiñas-Palacios, C.; Díaz-Bello, D. Energy Management Model for a Standalone Hybrid Microgrid through a Particle Swarm Optimization and Artificial Neural Networks Approach. *Energy Convers. Manag.* 2022, 267, 115920. [CrossRef]
- 115. Gamez Urias, M.E.; Sanchez, E.N.; Ricalde, L.J. Electrical Microgrid Optimization via a New Recurrent Neural Network. *IEEE Syst. J.* **2015**, *9*, 945–953. [CrossRef]
- 116. Qiu, X.; Nguyen, T.A.; Crow, M.L. Heterogeneous Energy Storage Optimization for Microgrids. *IEEE Trans. Smart Grid* 2016, 7, 1453–1461. [CrossRef]
- 117. Atia, R.; Yamada, N. Sizing and Analysis of Renewable Energy and Battery Systems in Residential Microgrids. *IEEE Trans. Smart Grid* 2016, 7, 1204–1213. [CrossRef]
- 118. Cao, M.; Xu, Q.; Cai, J.; Yand, B. Optimal sizing strategy for energy storage system considering correlated forecast uncertainties of dispatchable resources. *Int. J. Electr. Power Energy Syst.* **2019**, *108*, 336–346. [CrossRef]
- Okoye, C.O.; Solyalı, O. Optimal sizing of stand-alone photovoltaic systems in residential buildings. *Energy* 2017, 126, 573–584.
 [CrossRef]
- Akter, H.; Howlader, H.O.R.; Saber, A.Y.; Mandal, P.; Takahashi, H.; Senjyu, T. Optimal Sizing of Hybrid Microgrid in a Remote Island Considering Advanced Direct Load Control for Demand Response and Low Carbon Emission. *Energies* 2021, 14, 7599. [CrossRef]
- Malheiro, A.; Castro, P.M.; Lima, R.M.; Estanqueiro, A. Integrated sizing and scheduling of wind/PV/diesel/battery isolatedsystems. *Renew. Energy* 2015, 83, 646–657. [CrossRef]
- 122. Ren, H.; Wu, Q.; Gao, W.; Zhou, W. Optimal operation of a grid-connected hybrid PV/fuel cell/battery energy system for residential applications. *Energy* **2016**, *113*, 702–712. [CrossRef]
- 123. Chen, J.; Zhang, W.; Li, J.; Zhang, W.; Liu, Y.; Zhao, B.; Zhang, Y. Optimal Sizing for Grid-Tied Microgrids with Consideration of Joint Optimization of Planning and Operation. *IEEE Trans. Sustain. Energy* **2018**, *9*, 237–248. [CrossRef]
- 124. Kassab, F.A.; Celik, B.; Locment, F.; Sechilariu, M.; Liaquat, S.; Hansen, T.M. Optimal sizing and energy management of a microgrid: A joint MILP approach for minimization of energy cost and carbon emission. *Renew. Energy* 2024, 224, 120186. [CrossRef]
- 125. Agha Kassab, F.; Celik, B.; Locment, F.; Sechilariu, M.; Hansen, T.M. Combined Optimal Sizing and Energy Management of a DC Microgrid using MILP. In Proceedings of the 2023 IEEE Belgrade PowerTech, Belgrade, Serbia, 25–29 June 2023; pp. 1–6.
- 126. Agha Kassab, F.; Celik, B.; Locment, F.; Sechilariu, M. Optimal sizing of an isolated DC microgrid using multiobjective optimization with linear programming. In Proceedings of the Jeunes Chercheurs en Génie Électrique (JCGE22), Coisic, France, 14–17 June 2022.
- 127. Zhang, Y.; Campana, P.E.; Lundblad, A.; Yan, J. Comparative study of hydrogen storage and battery storage in grid connected photovoltaic system: Storage sizing and rule-based operation. *Appl. Energy* **2017**, 201, 397–411. [CrossRef]
- 128. Soykan, G.; Er, G.; Canakoglu, E. Optimal sizing of an isolated microgrid with electric vehicles using stochastic programming. *Sustain. Energy Grids Netw.* **2022**, *32*, 100850. [CrossRef]
- Li, B.; Roche, R.; Miraoui, A. Microgrid sizing with combined evolutionary algorithm and MILP unit commitment. *Appl. Energy* 2017, 188, 547–562. [CrossRef]
- 130. Boutros, F.; Doumiati, M.; Olivier, J.C.; Mougharbel, I.; Kanaan, H. New modelling approach for the optimal sizing of an islanded microgrid considering economic and environmental challenges. *Energy Convers. Manag.* **2023**, 277, 116636. [CrossRef]
- 131. Narayan, A.; Ponnambalam, K. Risk-averse stochastic programming approach for microgrid planning under uncertainty. *Renew. Energy* **2017**, *101*, 399–408. [CrossRef]

- 132. Balderrama, S.; Lombardi, F.; Riva, F.; Canedo, W.; Colombo, E.; Quoilin, S. A two-stage linear programming optimization framework for isolated hybrid microgrids in a rural context: The case study of the "El Espino" community. *Energy* **2019**, *188*, 116073. [CrossRef]
- 133. Stevanato, N.; Lombardi, F.; Colmbo, E.; Balderrama, S.; Quoilin, S. Two-Stage Stochastic Sizing of a Rural Micro-Grid Based on Stochastic Load Generation. In Proceedings of the 2019 IEEE Milan PowerTech, Milan, Italy, 23–27 June 2019; pp. 1–6.
- Schütz, T.; Hu, X.; Fuchs, M.; Müller, D. Optimal design of decentralized energy conversion systems for smart microgrids using decomposition methods. *Energy* 2018, 156, 250–263. [CrossRef]
- Liu, Y.; Beng Gooi, H.; Xin, H. Distributed energy management for the multi-microgrid system based on ADMM. In Proceedings of the 2017 IEEE Power & Energy Society General Meeting, Chicago, IL, USA, 16–20 July 2017; pp. 1–5.
- Rahmaniani, R.; Crainic, T.G.; Gendreau, M.; Rei, W. The Benders decomposition algorithm: A literature review. *Eur. J. Oper. Res.* 2017, 259, 801–817. [CrossRef]
- 137. Benders, J. Partitioning procedures for solving mixed-variables programming problems. *Numer. Math.* **1962**, *4*, 238–252. [CrossRef]
- 138. Geoffrion, A. Generalized Benders decomposition. J. Optim. Theory Appl. 1972, 10, 237–260. [CrossRef]
- 139. Ibrahim, A.; Dobre, O.A.; Ngatched, T.M.N.; Armada, A.G. Bender's Decomposition for Optimization Design Problems in Communication Networks. *IEEE Netw.* 2020, *34*, 232–239. [CrossRef]
- 140. Rahmaniani, R.; Ahmed, S.; Crainic, T.; Gendreau, M.; Rei, W. The Benders Dual Decomposition Method. *Oper. Res.* 2020, 68, 878–889. [CrossRef]
- 141. Masternak, C.; Meunier, S.; Brisset, S.; Reinbold, V. Microgrid sizing and energy management using Benders decomposition algorithm. *Sustain. Energy Grids Netw.* 2024, *38*, 101314. [CrossRef]
- 142. Yang, Y.; Pei, W.; Huo, Q.; Sun, J.; Xu, F. Coordinated planning method of multiple micro-grids and distribution network with flexible interconnection. *Appl. Energy* **2018**, 228, 2361–2374. [CrossRef]
- 143. Montoya-Bueno, S.; Muñoz-Hernandez, J.I.; Contreras, J.; Baringo, L. A Benders' Decomposition Approach for Renewable Generation Investment in Distribution Systems. *Energies* **2020**, *13*, 1225. [CrossRef]
- 144. Kim, T.H.; Shin, H.; Kwag, K.; Kim, W. A parallel multi-period optimal scheduling algorithm in microgrids with energy storage systems using decomposed inter-temporal constraints. *Energy* **2020**, 202, 117669. [CrossRef]
- 145. Chebabhi, A.; Tegani, I.; Benhamadouche, A.D.; Kraa, O. Optimal design and sizing of renewable energies in microgrids based on financial considerations a case study of Biskra, Algeria. *Energy Convers. Manag.* **2023**, 291, 117270. [CrossRef]
- 146. Ding, B.; Li, Z.; Xie, Y.; Chang, X.; Su, J.; Jin, X.; Sun, H. A CCP-based distributed cooperative operation strategy for multi-agent energy systems integrated with wind, solar, and buildings. *Appl. Energy* **2024**, *365*, 123275. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.